## University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Theses, Dissertations, and Student Research from Electrical & Computer Engineering

Electrical & Computer Engineering, Department of

Spring 1-2017

# Real-Time Internal Temperature Estimation and Health Monitoring for IGBT Modules

Ze Wang University of Nebraska - Lincoln, ze.wang.ken@huskers.unl.edu

Follow this and additional works at: http://digitalcommons.unl.edu/elecengtheses Part of the <u>Electrical and Electronics Commons</u>, <u>Electronic Devices and Semiconductor</u> <u>Manufacturing Commons</u>, and the <u>Power and Energy Commons</u>

Wang, Ze, "Real-Time Internal Temperature Estimation and Health Monitoring for IGBT Modules" (2017). *Theses, Dissertations, and Student Research from Electrical & Computer Engineering*. 74. http://digitalcommons.unl.edu/elecengtheses/74

This Article is brought to you for free and open access by the Electrical & Computer Engineering, Department of at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Theses, Dissertations, and Student Research from Electrical & Computer Engineering by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

## $Real-Time \ Internal \ Temperature \ Estimation \ and \ Health$

## MONITORING FOR IGBT MODULES

by

Ze Wang

### A DISSERTATION

Presented to the Faculty of

The Graduate College at the University of Nebraska

In Partial Fulfillment of the Requirements

For the Degree of Doctor of Philosophy

Major: Electrical Engineering

Under the Supervision of Professor Wei Qiao

Lincoln, Nebraska

January, 2017

## REAL-TIME INTERNAL TEMPERATURE ESTIMATION AND HEALTH MONITORING FOR IGBT MODULES

Ze Wang, Ph.D.

University of Nebraska, 2017

Advisor: Wei Qiao

Field experiences have demonstrated that power semiconductor devices, such as insulated-gate bipolar transistors (IGBTs), are among the most fragile components of power electronic converters. Thermomechanical stresses produced by temperature variations during operational and environmental loads are the major causes of IGBT degradation. As the devices are often operated under complex working conditions, temperature variations and the associated damage are difficult to predict during the converter design stage. A promising approach—online health monitoring and prognosis for power semiconductor devices—that can avoid device failure and effectively schedule maintenance has attracted much interest.

This dissertation research focused on real-time accurate internal temperature estimation and health condition monitoring for IGBT modules, where real-time means negligible latency. The objectives of this dissertation research were to: 1) develop lowcomputational-cost thermal models to accurately estimate important internal temperatures, e.g., junction temperature, in real time for better device overtemperature protection and lifetime prediction; and 2) invent a practical technique to monitor the major aging processes of IGBT modules for effective maintenance scheduling.

To achieve the first objective, this research developed a cost-effective, physicsbased thermal equivalent circuit model and a low-order digital filter-based thermal model for IGBT modules. Further, a thermal model adaptive to the solder crack of an IGBT module was developed. This model allows accurate junction temperature monitoring over a device's lifespan. Additionally, an accurate, frequency domain, transient temperature estimation method for weak points of IGBT modules was developed. The method can be used to determine the bottleneck weak point(s) of IGBT modules. The accurately estimated internal temperatures can be used for more effective operation management of power electronic converters and a more accurate estimation of a device's remaining useful lifetime (RUL).

To achieve the second objective, a new method of using two-dimensional (2D) case temperatures for real-time monitoring of typical weak points, i.e., bond wire and solder interface, of IGBT modules was invented. This method is based on the physics of failure of IGBT modules and can easily be implemented without complex circuitry. Compared with existing methods, the proposed method is simple to use and does not interrupt the operation of power electronic converters.

© Ze Wang 2017

#### Acknowledgements

Thanks are due foremost to my advisor, Professor Wei Qiao, for guiding me on how to conduct research and write technical papers. His guidance and encouragement throughout my study period have enhanced the entire learning and research experience.

Thanks also to Professors Jerry Hudgins, George Gogos, and Liyan Qu for their excellent suggestions in the areas of power electronics, heat transfer, and applications, respectively.

I would also like to acknowledge and thank my colleagues at the Power and Energy System Laboratory for their help with my research work. I wish to acknowledge an incomplete list of group members: Bo Tian, Chun Wei, Jianwu Zeng, Yue Zhao, Zhe Zhang, Cong Ma, Ting Dai, Dingguo Lu, Taesic Kim, Xiaohang Jin, Fa Chen, Fangzhou Cheng, Haibo Li, Haosen Wang, Yayu Peng, Lizhi Qu, Haixiang Zhang, and Xiaoqing Bai.

Finally, I would like to thank my family for their patient and gracious support.

Financial support from the following institutions/organizations is gratefully acknowledged:

- U.S. National Science Foundation (Grants ECCS-1308045 and ECCS-0901218 and CAREER Award ECCS-0954938)
- Office of Energy Efficiency and Renewable Energy (EERE), U.S. Department of Energy (Award DE-EE0006802)
- Power and Energy Systems Laboratory, Department of Electrical and Computer Engineering, University of Nebraska-Lincoln

### **Table of Contents**

Chapter 1. Introduction				
1.1	Background			
1.2	Causes of IGBT Module Failures			
1.3	Power Module Package			
1.4	Physics of Failure of IGBT Modules			
1.5	Juncti	on Temperature Estimation and Lifetime Prediction5	5	
1.6	Real-t	ime Health Monitoring and Prognostic Techniques	3	
1.7	Proble	em Statement9	)	
1.8 Dissertation Outline		tation Outline	)	
Chapter 2. Literature Review			2	
2.1	Therm	nal Models for IGBT Modules12	2	
	2.1.1	TEC Models 13	3	
	2.1.2	Analytical Models 15	5	
	2.1.3	Numerical Models	5	
	2.1.4	Digital Filter Models	7	
2.2	Condi	tion Monitoring Techniques for IGBT Modules	3	
	2.2.1	Bond Wire Aging Monitoring Using V <sub>CEon</sub>	3	
	2.2.2	Solder Aging Monitoring Using Temperature-Sensitive Electrical		
	Param	eters and R <sub>thjc</sub>	)	
	2.2.3	Other Electric Aging Precursor and Aging Monitoring Techniques . 21	L	
	2.2.4	Aging Monitoring Using Thermal Signal 22	2	

	2.3	Summary	vii 23
Char	ator 3	A Physics-Resol Improved Thermel Equivalent Circuit Model	24
Chaj	3.1	The Lumped Parameters of the Cauer-Type TEC	24
	3.2	Construction of the Improved Cauer-Type TEC	26
		3.2.1 Subdivision of the Layers in IGBT Modules	27
		3.2.2 Approximation of the Thermal Impedance of the Bulky Baseplate	: 29
	3.3	Simulation Validation	32
	3.4	Summary	36
Char Filte	pter 4 rs	. Real-Time Junction Temperature Estimation Using Low-Order Digi	tal 37
	4.1	An IGBT Module with Liquid Cooling	37
	4.2	IIR Digital Filter	38
	4.3	IGBT Thermal Modeling Using IIR Filters	39
	4.4	Simulation Validation	41
	4.5	Summary	47
Chap	pter 5	5. An Effective Heat Propagation Path-Based Online Adaptive Therma	l 1
Mod	el	EHPP-Based Thermal Behavior Characterization for IGBT Modules	<b>48</b>
	5.1	The EUDD Deced Online Adaptive Thermal Model for ICDT Modules	+)
	5.2	The EHPP-Based Online Adaptive Thermal Model for IGBT Modules	52
	5.3	Simulation Validation	55
	5.4	Summary	62
Chap	pter 6	. Real-Time Health Monitoring for IGBT Modules Using Case	<i>(</i> <b>)</b>
Tem	perat	Linear Thermal Behavior of IGBT Modules	<b>63</b> 64
	6.7	2D Case Temperature Daged Manitering of ICDT Madula A size	0 F
	0.2	2D Case Temperature-Based Monitoring of IGBT Module Aging	66

		viii
	6.2.1 Substrate Solder Aging	68
	6.2.2 Bond Wire and Emitter Metallization Aging	69
	6.2.3 Mixed Aging Mode of Bond Wires, Metallization, and Substrate	
	Solder 69	
	6.2.4 Other Major Aging Mechanisms	71
6.3	Implementation of the Proposed Method	71
6.4	Simulation Validation	75
6.5	Experimental Validation	81
	6.5.1 Validation of the Linear Thermal Behavior of the IGBT Module	82
	6.5.2 Solder Aging Monitoring	84
	6.5.3 Mixed Mode Monitoring of Aging	86
6.6	Summary	89
Chapter 7	. Frequency-Domain Transient Temperature Estimation and Aging	01
Analysis f	or Weak Points	91
7.1	Heat Equation for IGBT Modules	92
7.2	Frequency Domain Junction Temperature Estimation	92
7.3	Frequency Domain Aging Model Analysis	96
	7.3.1 Thermomechanical Stresses	96
	7.3.2 Aging Mode Analysis	97
7.4	Simulation Study	. 101
	7.4.1 Frequency Domain Junction Temperature Estimation	. 101
	7.4.2 Illustration of Temperature Reponses at Different Weak Points	. 106
	7.4.3 Frequency Domain Aging Mode Analysis	. 110

7	7.5	ix Summary111	
Chapte	er 8.	Conclusions, Contributions, and Recommendations for Future Research	
8	3.1	Conclusions and Contributions	
8	3.2	Recommendations for Future Research	
Appen	dix		
Refere	ences		
List of	List of Publications and Patent135		

## List of Figures

1.1	Mitsubishi CM400DY-12NF IGBT module [7]3
1.2	Cross-sectional view of a Mitsubishi NF series dual-switch IGBT module [7]4
1.3	Architecture of an IGBT module with the weak points circled by dashed lines 5
1.4	An example of a power cycling curve [16]7
2.1	TEC thermal models: (a) Foster-type and (b) Cauer-type 12
2.2	Schematic diagram of the in situ measurement circuitry in [45] 19
2.3	<i>V<sub>CEon</sub></i> measurement scheme (parallel diodes are not drawn) [46]20
2.4	Comparison of the different TSEPs [49]. (a) Voltage under a low current (diode,
IGBT,	MOSFET in OFF-state). (b) Threshold voltage. (c) Voltage under a high current.
(d) Ga	te-emitter voltage. (e) Saturation current. (f) Switching times
3.1	The architecture of an IGBT module with Cauer-TEC RC parameters
3.2	The conventional lumped Cauer-type TEC for an IGBT module25
3.3	Illustration of the temperature distributions in the layers of an IGBT module
estima	ted by the conventional lumped Cauer-type TEC and the actual continuous
temper	rature distribution of the IGBT module in the steady-state condition
3.4	The exact description of the thermal behavior of the baseplate. (a) The sketch of
the ge	ometry and heat spreading and (b) the distributed Cauer-type TEC model for the
basepl	ate
3.5	The model of a CM400-12NF IGBT module with a cold plate built in Autodesk
Simula	ation Multiphysics

	xi
3.6	Comparison of the transient thermal impedances from junction to case obtained
from t	he conventional Cauer-type TEC model, the proposed Cauer-type TEC model, and
the FE	A model
3.7	Comparison of junction temperature estimation by the three models when the
inverte	er was operated at a constant operating condition: (a) $\Delta T_{thjc}(t)$ ; and (b) errors with
respec	t to the FEA model
4.1	An IGBT module (only one chip is shown) cooled by a cold plate
4.2	Comparison of the thermal impedances modeled by the 6th-order Foster-type
TECs,	the proposed 5th-order IIR filters, and the FEA method: (a) $Z_{thjaQQ}(t)$ and
$(b)Z_{thjo}$	<sub>aQD</sub> (t)
4.3	Comparison $T_j(t)$ of the IGBT estimated by the three methods when the inverter
is oper	rated at a constant operating condition: (a) junction temperature and (b) error with
respec	t to the FEA method
4.4	Total power loss profile of the IGBT chips in one switch of the IGBT module in
a trans	ient load condition
4.5	Comparison $T_j(t)$ estimated by the three methods when the inverter is operated
with a	transient load: (a) junction temperature; and (b) error at peaks and valleys with
respec	t to the FEA method
5.1	The EHPP in a healthy IGBT module
5.2	The EHPP in an IGBT module when cracks developed from the edge to the center
in the	substrate solder layer
5.3	IHCP using case temperature to estimate the effective heat flux area

5.4	The approximated EHPP for an IGBT module with substrate solder cracks 53
5.5	Schematic of the proposed online adaptive TEC model
5.6	The model of one-sixth of a CM1400 IGBT module built in ADINA
5.7	Baseplate side view of the IGBT module
5.8	EHPPs of the side dies and the center die (separated by dashed lines)
5.9	Power loss profiles of constant load and dynamic load cases for one IGBT chip.58
5.10	Changes in nonuniformity of the case temperature distribution and maximum
junctio	on temperature in the three substrate solder crack cases in constant and dynamic
load co	onditions
5.11	Comparison of $R_{thja,side}$ calculated by the FEA method and the proposed model
	60
5.12	Comparison of different models for the cases of 3.5-mm solder cracks and 5-mm
solder	cracks under the dynamic load condition: (a) junction temperature (3.5-mm solder
crack)	; (b) junction temperature (5-mm solder crack); (c) relative error (3.5-mm solder
crack)	; and (d) relative error (5-mm solder crack)
6.1	An IGBT module cooled by a cold plate with selected points for 2D case
temper	rature measurements
6.2	The placement of thermal sensors on the bottom surface of the baseplate of a
three-p	phase inverter module having six IGBT switches
6.3	The flowchart of the implementation of the aging monitoring method using two

6.4	The test IGBT module with the positions of the two thermal sensors placed on the
botton	n surface of the baseplate (top view from the upper side of the module)
6.5	Evolution of parameters during the mixed mode aging process of the IGBT
modul	e: (a) $T_{cchip}$ , $T_{cside}$ , and $T_j$ ; (b) $P_{loss}(T_j)$ ; and (c) $R_{thjc}$
6.6	Evolution of the estimated parameters during the mixed mode aging process of
the IG	BT module: (a) $\hat{R}_{eqCchip}$ and $\hat{R}_{eqCside}$ ; (b) k
6.7	Calculated $\alpha_P$ during the aging process of the IGBT module
6.8	Experimental setup for validation of the proposed method: (a) the schematic and
(b) the	actual setup
6.9	The values of $R_{eqCchip}$ , $R_{eqCside}$ and $k_{cs}$ at different operating conditions
6.10	Evolution of parameters during thermal cycling of the IGBT module: (a)
measu	red $T_{cchip}$ , $T_{cside}$ , and $T_j$ ; (b) $R_{thjc}$
6.11	Evolution of $k_{cs}$ during the thermal cycling test of the IGBT module
6.12	Evolution of $V_{CEon}$ during the power cycling test of the IGBT module
6.13	Evolution of parameters during the power cycling test of the IGBT module: (a)
$\alpha_P$ and	(b) <i>k<sub>cs</sub></i>
7.1	The flowchart of the proposed method
7.2	Selected points and heat flows inside an IGBT module
7.3	Temperature information of $P_{jm}$ and $P_{se}$ under two different loading conditions

viv			
7.4 The transient thermal impedance responses from the junction of the IGBT chip to			
the case and the corresponding LTI system responses: (a) $Z_{thjcQQ}(t)$ ; (b) $Z_{thjcQD}(t)$ ; (c)			
$H_{QQ}(t)$ ; and (d) $H_{QD}(t)$			
7.5 Comparison of $T_j(t)$ estimated by the three methods when the inverter was			
operated with a transient load: (a) junction temperature and (b) error at peaks and valleys			
with respect to the FEA method			
7.6 Comparison of $T_j(t)$ estimated by the proposed method and the FEA method 106			
7.7 FEA model of the general IGBT module 107			
7.8 FEA simulation results for a step-change input power. (a) Normalized transient			
thermal impedances from the selected points to ambient and (b) frequency responses of			
the temperatures at $P_{jm}$ , $P_{ce}$ , and $P_{se}$			
7.9 Comparison of the proposed method with the FEA method. (a) The power loss			
and the transient temperatures estimated by the two methods at (b) $P_{jm}$ , (c) $P_{ce}$ , and (d)			
<i>P<sub>se</sub>.</i>			
7.10 Speeds of HEV, frequency spectra of heat generation, and temperatures of $P_{se}$ for			
loading (a) and loading (b) 111			
A.1 The temperature distribution approximation error of the <i>i</i> th layer subdivided into			
<i>N</i> sublayers			

### List of Tables

2.1	Analogy between thermal and electrical variables
2.2	Comparison of existing thermal models for IGBT modules17
2.3	Comparison of existing condition monitoring methods for IGBT modules
3.1	Typical layer thicknesses and thermal parameters (without subdivision) in a high-
pow	er IGBT module
6.1	Changes of key parameters for different aging mechanisms71
6.2	Lookup table for solder aging monitoring
6.3	The measured temperatures at different operation conditions
6.4	Comparison of aging monitoring methods for IGBT modules

## List of Acronyms/Initialisms/Nomenclature

## A. Acronyms/Initialisms

1D	one-dimensional
2D	two-dimensional
3D	three-dimensional
AlN	aluminum nitride
CTE	coefficient of thermal expansion
DBC	direct bonded copper
DC	direct current
DFT	discrete Fourier transform
EHPP	effective heat propagation path
FEA	finite element analysis
FIR	finite impulse response
FFT	faster Fourier transform
FWD	free-wheeling diode
HEV	hybrid electric vehicle
IC	integrated circuit
IGBT	insulated-gate bipolar transistors
IHCP	inverse heat conduction problem
IIR	infinite impulse response
LESIT	Leistungselektronik, Systemtechnik und
	Informationstechnologie

LTI	linear time invariant
MOSFET	metal-oxide semiconductor field-effect transistor
NI	National Instruments
PRBS	pseudo random binary sequence
PF	power factor
PSU	power supply unit
RC	resistor-capacitor
RUL	remaining useful lifetime
TEC	thermal equivalent circuit
ТО	transistor outline
TSEP	temperature-sensitive electrical parameter

α	thermal diffusivity
( <i>Aadj</i>	heat spreading angle in the substrate copper layer needs to
	be adjusted during substrate solder cracks
$\Delta a_{CTE}$	the difference between the CTEs of two different materials
$lpha_f$	a constant in lifetime model (1.4)
αр	the ratio between actual power loss $P_{loss}(T_J)$ and estimated
	power loss $\hat{P}_{loss}(\hat{T}_J)$
θ	heat spreading angle in degree
Emismatch	the strain at the interface between two different materials
$\mathcal{E}_p$	plastic strain
$\mathcal{E}_{f}^{\prime}$	fatigue ductility coefficient
$\mathcal{E}_e$	elastic strain
$\sigma_a$	alternating stress
$\sigma_f'$	fatigue strength coefficient
$ au_{iF}$	time constant of the $i_{Foster}$ th RC pair
A(z)	horizontal cross-section area at the distance $z$
$A_c$	area of the EHPP through the bottom surface of the ceramic
	layer
$A_f$	a constant in lifetime model (1.4)
A <sub>hot</sub>	baseplate hot area
$A_{IIR}(z)$	coefficient in an IIR filter

	xix
Ar	remained non-cracking area in the substrate solder layer
$b_f$	fatigue strength exponent
$B_{IIR}(z)$	coefficient in an IIR filter
$CE_{lc}$	lumped-capacitance approximation error
$CE_{lc,i,sub}$	lumped-capacitance approximation error of each sublayer
Cf	fatigue ductility exponent
$C_{f}$	normalized total damage in one operation cycle
C <sub>th</sub>	thermal capacitance
$C_{th}(z)$	distributed heat capacitance at vertical distance $z$
$C_{th,i}$	lumped thermal capacitance of the <i>i</i> th layer
$C_{th,iF}$	thermal capacitance of the $i_{Foster}$ th RC pair
C <sub>th,tot</sub>	total thermal capacitance of the IGBT module
$C_{V}$	volumetric heat capacity
Cvbp	volumetric specific heat of the copper baseplate
$d_{bp}$	thickness of the baseplate
$d_i$	thickness of the <i>i</i> th layer
$d_l$	thickness of a layer
d <sub>pattern</sub>	thickness of the substrate copper pattern layer
Ε	Young's modulus
$E_a$	activation energy
fsw	switching frequency
F <sup>-1</sup>	inverse Fourier transform

H(z)	transfer function of an IIR filter
H <sub>nm</sub>	frequency response of the temperature rise from the <i>n</i> th
	junction to case excited by the <i>m</i> th source power loss $P_m$
$H_{nn}(z)$	IIR filter charactering the self-heating of the <i>n</i> th chip
$H_{nm}(z)$	IIR filter charactering the relationship between junction
	temperature of the <i>n</i> th chip, $T_{jn}$ , and power loss of the <i>m</i> th
	chip
$h_{pj}(t)$	high-order LTI system characterizing relationship between
	junction-to-case temperature difference $\Delta T_{jc}(t)$ and power
	loss $P_{loss}(t)$
$h_{pj}[k]$	discrete-time form of $h_{pj}(t)$
$H_{pj}(e^{j\omega})$	frequency response of $h_{pj}(t)$
$H_{pj}(\Omega)$	frequency response of $h_{pj}[k]$
$h_{QD}(t)$	time derivatives of thermal impedance $Z_{thjcQD}(t)$
$h_{QQ}(t)$	time derivatives of thermal impedance $Z_{thjcQQ}(t)$
i	index for the layers from top to bottom in IGBT modules
Ic	collector current
<i>i</i> <sub>Cf</sub>	index for different junction temperature swing $\Delta T_j$ in one
	operation cycle
<i>i</i> <sub>F</sub>	index of RC pairs in a Foster-type TEC
J	matrix of ones
k	thermal conductivity

	xxi
k <sub>B</sub>	Boltzmann constant
$k_{bp}$	thermal conductivity of the copper baseplate
<i>k</i> <sub>cs</sub>	ratio between equivalent thermal resistance $R_{eqCchip}$ and
	equivalent thermal resistance $R_{eqCside}$
<i>k</i> <sub>dt</sub>	index for discrete-time sequence
k <sub>hc</sub>	the ratio between equivalent thermal resistance $R_{eqCc}$ and
	equivalent thermal resistance $R_{eqCc}$
$K_{sp}$	number of the time-domain data samples of the power
	losses
n <sub>f,i</sub>	cycle occurrence of the $i_{Cf}$ th $\Delta T_j$
Ncf	number of different junction temperature cycle amplitudes
$N_{f}$	number of cycles to failure
N <sub>f,i</sub>	number of cycles to failure for the $i_{Cf}$ th $\Delta T_j$
Nı	order of a LTI system
N <sub>rc</sub>	order of an RC TEC
Nsub	number of sublayers
P <sub>ce</sub>	point at the free edge of the chip solder layer
P <sub>Closs</sub>	conduction loss
<i>P</i> <sub>flux</sub>	heat flux
Ploss	total power loss
$P_{loss}[k]$	discrete time power loss
$P_{loss}(\Omega)$	discrete Fourier transform (DFT) of $P_{loss}[k]$

	xxii
$\hat{P}_{loss}(\hat{T}_j)$	calculated power loss at estimated junction temperature $\hat{T}_j$
	using healthy device parameters
$P_{jm}$	point at the middle of junction
$P_n$	power loss of the <i>n</i> th chip
Pse	point at the free edge of the substrate solder
PSWloss	switching loss
q(z)	heat flow
$Q_i$	amount of heat stored in the <i>i</i> th layer
$\mathbf{R}_{eqC}$	matrix of the equivalent thermal resistances from the points
	of interest on the 2D bottom surface of the baseplate to
	ambient
$R_{eqCc}$	an element in the matrix $\mathbf{R}_{eqC}$ for a cold spot
$R_{eqCh}$	an element in the matrix $\mathbf{R}_{eqC}$ for a hot spot
$R_{eqcp}$	equivalent thermal resistance characterizing the
	temperature rise from ambient to any point $p$
$R_{eqCchip}$	equivalent thermal resistances between the selected point
	right underneath chip and ambient
$R_{eqCside}$	equivalent thermal resistances between the selected point
	away from the hot spot and ambient
$\hat{R}_{eqCchip}$	calculated equivalent thermal resistances between the
	selected point right underneath chip and ambient

	xxiii
$\hat{R}_{eqCside}$	calculated equivalent thermal resistances between the
	selected point away from the hot spot and ambient
<i>R</i> <sub>th</sub>	thermal resistance
$R_{th}(z)$	distributed thermal resistance at vertical distance $z$
$R_{th,iF}$	thermal resistance of the $i_{Foster}$ th RC pair
<b>R</b> <sub>thca</sub>	thermal resistance from case to ambient
<b>R</b> <sub>thja</sub>	thermal resistance from junction to ambient
<b>R</b> <sub>thjc</sub>	thermal resistance from junction to case
<b>R</b> <sub>thja,side</sub>	thermal resistance from side chip junction to ambient
S	Laplace variable
t	continuous time
$ au_{iF}$	corresponding time constant
T <sub>C</sub>	case temperature
T(z)	temperature at z
$\Delta T_{arepsilon}$	the temperature change from the temperature where the
	strain is zero
$T_a$	ambient temperature
$\mathbf{T}_{C}$	steady-state 2D case temperatures
$T_{cchip}$	case temperature at the location right beneath IGBT chip
$T_{cool}$	coolant temperature
$T_{cside}$	case temperature near the edge of the $A_{hot}$
$T_j$	junction temperature

	vviu
$\hat{T}_j$	estimated junction temperature
$\Delta T_j$	junction temperature swing
$\Delta T_{jc}$	junction-to-case temperature difference
$T_{jmax,side}$	maximum junction temperature of the side chip
$T_{jn}$	junction temperature of the <i>n</i> th chip
$T_m$	average junction temperature
$T_s$	sampling period
<i>u</i> ( <i>t</i> )	input of a target system
VCEon	ON-state collector-emitter voltage
<i>y</i> ( <i>t</i> )	output of a target system
Z	vertical distance from the top surface of the layer to the
	point of interest
$Z_{nm}(t)$	equivalent transient thermal impedance from the <i>n</i> th
	junction to case when only the <i>m</i> th source is powered
$Z_{th,bp}$	transient thermal impedance of the baseplate
$Z_{thjaQQ}(t)$	the self-heating thermal impedance from the IGBT junction
	to ambient
$Z_{thjaQD}(t)$	cross-heating thermal impedance from the IGBT junction
	to ambient
Z <sub>thjc</sub>	transient thermal impedance from junction to case
$Z_{thjcQQ}(t)$	transient thermal impedances from the junction of the
	IGBT chip to case when the IGBT chip is powered alone

$Z_{thjcQD}(t)$	transient thermal impedances from the junction of the	
	IGBT chip to case when the parallel FWD is powered	alone

## **Chapter 1.** Introduction

### 1.1 Background

The reliability of power electronic converters (or more simply called power converters) in many applications, such as high-power traction and renewable energy generation, has attracted considerable attention in recent years. Power converters tend to experience a higher failure rate than other subsystems in these applications. For example, it was reported that the power electronic converter is the top-ranked subsystem contributing to the overall failure rates and downtime of wind turbines [1], [2]. As most power converters are not designed with redundancy, any faults occurring at the component or subsystem level would cause the shutdown of the whole power converter system. The unexpected interruptions cause safety concerns and result in high operation and maintenance costs, especially for critical applications, such as automobile, backup power source, aircraft power system, etc.

Power semiconductor devices, such as insulated-gate bipolar transistors (IGBTs) and metal-oxide semiconductor field-effect transistors (MOSFETs), were rated as the most fragile components of a power converter in a recent industrial survey [3]. The reliability research in power electronics is now switching from an empirical approach based on previously observed data to a physics-of-failure approach which is more focused on the root causes behind the failures [4], [5]. The physics-of-failure approach can model potential failure mechanisms and integrate investigative results into the design process to improve the robustness of a device. During operation, the physics-of-failure-

based online monitoring systems of power converters can collect useful information, such as junction temperatures for the remaining useful lifetime (RUL) prediction, or directly detect the aging conditions of the devices. The operation of power converters can be more properly managed, and predictive maintenance can be automatically scheduled before components wear out.

### **1.2 Causes of IGBT Module Failures**

Failures of IGBT modules can be roughly classified into two categories: random failures and wear-out failures [9]. Random failures are usually caused by external accidental events, such as particle radiation, voltage transients, and damage by service actions leading to momentary overstress [6]. These failures are not related to the length of service or the age of the device. The wear-out failures are attributed to the accumulation of incremental physical damage under the operating load (stress) conditions, altering the device properties beyond the functional limit [6].

Mitsubishi Electric Corporation, a major IGBT manufacturer, has summarized the causes of IGBT module failures based on an analysis of the failed devices that are returned to their factory. According to the manufacturer's report [6], although problems may be inherent to the device itself or the manner in which it was used, the operational lifetime of the device, however, is dependent on the internal wire bonding and thermal fatigue between the insulation baseplate and the module baseplate.

This dissertation focuses on the major wear-out failures of the bond wire and solder interface caused by thermal fatigue.

### **1.3 Power Module Package**

As the wear-out failures are related to the IGBT package, the power module will be introduced in this section. Power modules are the prevalent types of packages in power electronic applications, dominating in the range of more than 10 A for blocking voltages of 1200 V and above [17]. Power modules may also be integrated with several single functions, e.g., dual-switch, half-bridge leg, often with paralleling of chips for each switch. Fig. 1.1 shows a Mitsubishi IGBT module with two switches in a half-bridge configuration. This module has three main terminals (electrodes) which connect to a voltage source and a load. Each switch is controlled by an external signal that is applied to a pair of gate/emitter auxiliary pins. The structure of this module is illustrated in Fig. 1.2. Silicon chips are placed on a ceramic substrate. The terminals are connected to silicon chips via aluminum wires and the upper copper foil. Silicone gel provides encapsulation and can absorb some damaging vibration.



Fig. 1.1: A Mitsubishi IGBT module [7].



Fig. 1.2: Cross-sectional view of the structure of a Mitsubishi IGBT module [7].

### 1.4 Physics of Failure of IGBT Modules

In this section, the architecture of IGBT modules will be discussed in detail to explain the physics of failure. Typically, an IGBT module has a complex multilayered architecture, as illustrated in Fig. 1.3. The transistor chip is soldered to the upper copper surface of a ceramic "direct bonded copper" (DBC) substrate. The top side contacts of the silicon chips are connected via wire bonds. Trenches of the upper copper pattern layer of the DBC substrate form current tracks. The substrate is attached to a baseplate by solder. The baseplate is usually mounted onto a heat sink or cold plate with thermal interfacial materials applied on the contact surface. It should be noted that Fig. 1.3 is only for illustration purposes and does not reflect the real geometry of the IGBT module.

The failure of IGBTs can be the result of external accidental events, e.g., an abnormal control signal or bus voltage rise, or the wearing out or aging of the packaging. The lifetime of the device is considered to be limited by aging of the device [6], which is mainly caused by the cyclic thermomechanical stresses generated during temperature excursions due to the mismatch in coefficients of thermal expansion (CTE) between the



Fig. 1.3: Architecture of an IGBT module with the weak points circled by dashed lines.

materials used in the packaging. The major aging mechanisms frequently observed in IGBT modules are bond-wire liftoff, reconstruction of metallization, and solder aging [10], . Because the CTEs of copper and aluminum are much larger than those of silicon and ceramic, the solder interfaces and the interface between bond wire and chip are more severely stressed when the module undergoes temperature fluctuations. The weak points are shown in Fig. 1.3.

The reliability of power converters can be improved by choosing IGBT modules with more rugged packaging designs, including wire connection optimization [13] and sintered silver for die attachment [14]. However, the trade-off between robustness and cost is always a concern; and the expected lifetime of IGBTs can hardly be guaranteed, especially when the power converters operate under complex work conditions.

### **1.5 Junction Temperature Estimation and Lifetime Prediction**

Information on junction temperature  $T_j$  is crucial for operational management of IGBT modules in real-world applications. Accurate junction temperature estimation can be used to prevent overtemperature [15] and predict the RUL [17] for an IGBT module. In practice, junction temperature cannot be directly measured. Typically, it is estimated

by using an electrothermal model, which consists of power loss estimation, a power module thermal model, a cooling system thermal model, and a reference temperature, such as ambient temperature  $T_a$  or coolant temperature  $T_{cool}$ .

The total power loss  $P_{loss}$  of the device includes conduction loss  $P_{Closs}$  and switching loss  $P_{SWloss}$ :

$$P_{loss} = P_{Closs} + P_{SWloss} \tag{1.1}$$

 $P_{Closs}$  and  $P_{SWlos}$  can be calculated using the equations in [20] and the device datasheet parameters, which may have temperature dependence.

Two important concepts for thermal modeling of IGBT modules are the thermal resistance from junction to case  $R_{thjc}$  and transient thermal impedance from junction to case  $Z_{thjc}$ .  $R_{thjc}$  is a steady-state value given by:

$$R_{thjc} = \frac{T_j - T_c}{P_{loss}} \tag{1.2}$$

where  $T_c$  is the case temperature of the module measured at a specified point on the outside surface of the package. The transient thermal impedance from junction to case  $Z_{thjc}$  is calculated based on the measured time-varying  $T_j$  (*t*) and  $T_c$  (*t*) in the device heating phase when a constant IGBT power loss applies.

$$Z_{thjc} = \frac{T_j(t) - T_c(t)}{P_{loss}}$$
(1.3)

According to (1.2), if the average junction temperature is of concern, it can be calculated by using  $R_{thjc}$ , the calculated average power loss, and the average case temperature. However, to calculate transient  $T_j(t)$ , a dynamic thermal model should be used. The most commonly used real-time thermal models are thermal equivalent circuits (TECs), such as the Foster- or Cauer-type TEC. The TECs, however, cannot provide an accurate junction temperature estimation in fast-varying loading conditions due to their unsatisfactory transient responses. More details of the TEC models and other thermal models will be presented in Chapter 2.

The advent of failure of IGBT modules can be predicted by calculating the RUL, which typically uses a lifetime model and information such as IGBT junction temperature. The lifetime of a power IGBT module in the number of cycles to failure can be calculated by using the Leistungselektronik, Systemtechnik und Informationstechnologie (LESIT) model:

$$N_f = A_f \cdot \Delta T_j^{\alpha_f} \cdot \exp(\frac{E_a}{k_B \cdot T_m})$$
(1.4)

where  $\Delta T_j$  is the temperature swing of junction,  $T_m$  is the average junction temperature,  $k_B$  is the Boltzmann constant,  $E_a$  is the activation energy, and the constants  $A_f$  and  $\alpha_f$  are determined by experiments. For example, the constants in the model (1.4) for a specific device can be obtained by fitting the device's power cycling curve. An example of a power cycling curve is give in Fig. 1.4. The horizontal axis is the amplitude of  $\Delta T_j$  in



Fig. 1.4: An example of a power cycling curve [16].

each power cycle, and the vertical axis is the number of cycles that the devices fail at an accumulated failure rate, which is typically 20%.

If the typical operation cycles of applications are known, it is possible to calculate the expectation for the lifetime of an IGBT module under these conditions. In real-world applications, many  $\Delta T_j$  with different amplitudes usually exist in an operation cycle of a power converter. Cycle counting is used to summarize lengthy, irregular  $T_j$ -versus-time histories by providing the number of cycles of various  $\Delta T_j$  amplitudes occurred. Then, the lifetime of the device is typically calculated by using the Palmgren-Miner's rule, which assumes damage accumulation is independent of stress level. If there are  $N_{cf}$  different  $\Delta T_j$ amplitudes, the normalized total damage in one operation cycle is:

$$C_f = \sum_{i_{cf}=1}^{N_{cf}} \frac{n_{f,i_{cf}}}{n_{f,i_{cf}}}$$
(1.5)

where  $i_{Cf}$  is the index of different  $\Delta T_j$  amplitudes,  $N_{f,iCf}$  is the number of cycles to failure for the  $i_{Cf}$ th  $\Delta T_j$  amplitude, and  $n_{f,iCf}$  is the number of the cycles occurred with the  $i_{Cf}$ th  $\Delta T_j$  amplitude. The device is considered a failure after  $1/C_f$  operation cycles.

However, an RUL prediction for IGBT modules is still not accurate in practice. One cause is that the accuracy of the conventional TEC model is often unsatisfied and degrades during the aging process (such as solder cracking), which results in an inaccurate estimation of  $T_j$ , especially for power converters working under complex conditions.

### **1.6 Real-time Health Monitoring and Prognostic Techniques**

Real-time monitoring and prognosis of IGBT modules is considered to be a promising approach to avoid the failure of IGBTs and improve the long-term reliability of power converters. The aging of bond wires and metallization in an IGBT module results in an increase of the ON-state collector-emitter voltage  $V_{CEon}$ ; while fatigue of solder interfaces increases  $R_{thjc}$ . The commonly used criteria indicating the dangerous health status of IGBT modules are: (1)  $V_{CEon}$  has increased by 10~20%, or (2)  $R_{thjc}$  has increased by more than 20%. Various techniques for measuring  $V_{CEon}$  and  $T_j$  (for  $R_{thjc}$ calculation) were proposed and tested in a lab environment. However, the changes in  $V_{CEon}$  and  $R_{thjc}$  are difficult to measure in practice. The existing aging precursors [18] and aging detection techniques [19] for IGBT modules are still premature and/or impractical for real-world applications. These precursors and techniques will be discussed in detail in Chapter 2.

### **1.7 Problem Statement**

As stated in the previous sections, the long-term reliability of power converters can be more cost-effectively improved by real-time overtemperature protection, online RUL prediction, and real-time health monitoring of IGBT modules. However, the existing thermal models for IGBTs are not accurate enough to estimate transient junction temperature for overtemperature protection and RUL prediction. And, there is a gap between the existing aging precursors and real-time monitoring techniques and practical implementation in real-world applications.

The objectives of this research were to: 1) develop cost-effective thermal models to accurately estimate important internal temperatures, e.g., junction temperature, of IGBT modules in real time for better device overtemperature protection and RUL prediction; and 2) invent a practical technique to monitor the major aging processes of IGBT modules in real time for effective maintenance scheduling.

### **1.8 Dissertation Outline**

This dissertation is organized as follows.

Chapter 2 provides a literature review of electrothermal models and condition monitoring techniques for IGBT modules.

Chapter 3 presents a method for designing a physics-based improved Cauer-type TEC thermal model. This model greatly improves the accuracy of the conventional Cauer-type TEC model for the real-time transient junction temperature estimation of IGBT modules with a slightly increased order only.

Chapter 4 describes the characterization of the thermal behavior of an IGBT module using low-order infinite impulse response (IIR) digital filters. The order of the IIR filter thermal model is low because it takes advantage of using the junction temperature as a feedback. Another advantage is that the IIR filter thermal model can be easily implemented in a digital processing system.

Chapter 5 presents a new concept for an effective heat propagation path (EHPP) to interpret the effect of the substrate solder crack on the heat propagation inside an IGBT module. Based on EHPP, a thermal model adaptive to the solder health status can be built; and its parameters can be updated online. This allows accurate real-time junction temperature monitoring over a device's lifespan and a more accurate online prediction of a device's RUL.

Chapter 6 presents a two-dimensional case temperature-based, real-time aging monitoring method for multiple aging processes of IGBT modules in power converters. This approach is based on the physics of failure of IGBT modules and interprets complex aging processes via the changes in the device thermal behaviors. Compared to existing
methods, the proposed method does not interrupt the operation of power converters and can be easily implemented without complex circuitry.

Chapter 7 presents a frequency-domain thermal analysis method to efficiently estimate the temperature swings at the weak points of IGBT modules. By examining the frequency spectra of the power losses of an IGBT module at a loading condition and the frequency responses of the temperatures at weak points, the information on bottleneck weak points under the specific operating conditions can be obtained.

Chapter 8 summarizes the contributions of this doctoral dissertation research. Recommendations for future research are also presented.

# **Chapter 2.** Literature Review

This chapter provides a comprehensive review of existing thermal models as well as the state-of-the-art aging monitoring techniques for IGBT modules.

# 2.1 Thermal Models for IGBT Modules

The junction temperature of an IGBT module is usually estimated online by using a thermal model, which characterizes the dynamical thermal behavior of the IGBT module. These thermal models mainly include the conventional thermal equivalent circuit (TEC) models, analytical models, numerical models obtained by methods such as Finite Element Analysis (FEA), and digital filter models.



Fig. 2.1: TEC thermal models: (a) Foster-type and (b) Cauer-type.

#### 2.1.1 TEC Models

TECs are the most commonly used models for thermal modeling of IGBT modules, for example [21]-[24]. The TECs have two basic topologies: the Foster-type TECs and the Cauer-type TECs, as shown in Fig. 2.1.

The Foster-type TEC is represented by a certain number of thermal resistorcapacitor (RC) pairs. The values of the thermal RC pairs are typically determined by fitting the sum of the first-order transfer functions of all the thermal RC pairs of the Foster-type TEC to the  $Z_{thic}(t)$  of an IGBT module as follows.

$$Z_{thjc}(t) = \sum_{i_F=1}^{N_{rc}} R_{th,i_F}(1 - e^{-\frac{t}{\tau_{i_F}}})$$
(2.1)

where *t* is time,  $N_{rc}$  is the order of the RC circuit (i.e., the number of thermal RC pairs),  $i_F$ is the index of thermal RC pairs, and  $R_{th,i_F}$ ,  $C_{th,i_F}$ , and  $\tau_{i_F}$  (= $R_{th,i_F} \cdot C_{th,i_F}$ ) are the thermal resistance, thermal capacitance, and time constant of the  $i_F$ th thermal RC pair, respectively. The Foster-type TEC has limitations in estimating fast transient temperature changes of the IGBT module [25]. When  $t/\tau_{i_F} \rightarrow 0$ ,

$$Z_{thjc}(t) = \sum_{i_F=1}^{N_{rc}} R_{th,i_F}(1 - e^{-\frac{t}{\tau_{i_F}}}) \approx (\sum_{i=1}^{N} \frac{R_{th,i_F}}{\tau_{i_F}}) \cdot t$$
(2.2)

Equation (2.2) indicates that the Foster-type TEC can only accurately model the special case in which  $Z_{thjc}(t)$  has a slope of one in the log-log coordinates because the power of t is about one. Otherwise, a relatively large error will occur, which makes accurate junction temperature estimation impossible under fast-changing power loss conditions.

There is an analogy between the heat diffusion equation and the differential equation for a transmission line with distributed line resistance and shunt capacitance [26]. The analogy between thermal and electric variables is shown in Table 2.1. A Cauer-type TEC of an IGBT module is constructed based on the analogy and the geometry of the module, where the nodes, thermal resistances, and thermal capacitances have physical meanings.

The parameters of the Cauer-type TEC can be adjusted by considering the temperature dependence of the materials' properties [27], [28], leading to an adaptive thermal model. To accurately characterize the thermal behavior of an IGBT module, a high-order Cauer-type TEC with distributed thermal resistance and capacitance is typically required. On the other hand, a low-order, high-accuracy, Cauer-type TEC model is always desired for online or real-time applications.

For a power semiconductor device with relatively simple packaging, e.g., the "transistor outline" (TO) package, a low-order Cauer-type TEC can be obtained based on the time constant spectra calculated from the thermal step response of the device [29], [30]. However, that method assumes that the thermal behavior of the packaging can be represented by the sum of a few first-order systems; and their time constants are widely

	Variables		
	Thermal	Electrical	
Through variable	heat flow rate, $q$	current, I	
Across variable	temperature, T	voltage, V	
Resistance	thermal resistance, $R_{th}$	resistance, R	
Storage	thermal capacity, $C_{th}$	capacitance, C	

Table 2.1: Analogy between thermal and electrical variables.

separated [31]. Therefore, that method may not work for the IGBT modules with a relatively complex architecture.

The authors in [32] introduced the Elmore delay formula to model the propagation delay times of the heat flux in the layers of an IGBT module. Then, the time constants and thermal capacitances of the layers can be calculated using the calculated delay times and the measured thermal resistances of the layers. However, the application of the Elmore delay formula to the parameter calculation of the Cauer-type TEC is questionable because the Elmore delay formula is typically used for calculating the RC interconnect delays between voltage drivers and ideal (infinite impedance) loads.

The conventional Cauer-type TEC model neglects the temperature difference in each layer of the IGBT module and models each layer by a thermal resistance and a lumped thermal capacitance [30], [33]. The lumped thermal capacitance modeling is oversimplified for bulky layers, e.g., the ceramic layer and baseplate layer, which reduces the transient performance of the conventional Cauer-type TEC. Therefore, the accuracy of the dynamic junction temperature of the IGBT module estimated by using the conventional Cauer-type TECs is unsatisfactory.

#### 2.1.2 Analytical Models

Analytical models, such as [34], [35], are usually obtained based on series approximation techniques to solve heat transfer equations, which are mathematically complex even if the boundary conditions are simplified. Moreover, these models are often subject to problems such as series truncation and convergence [36].

For example, an exact solution to heat conduction in a simple planar plate with power law and polynomial heat flux input was presented in [37]. Green's functions were used and resulted in slowly converging summation, which had to be replaced with closedform polynomial expressions. The final solution was very complex. In [38], for heat conduction in a two-layer composite slab, a semi-analytical procedure was developed to obtain a relatively "simple" solution. The problem was simplified by analyzing the different transition times of the two layers of the slab. However, that method had to deal with the irregularly spaced eigenvalues of an eigenproblem with coefficients having the step change at an interior point. The temperature profile of the homogeneous layer of a two-layer material was found very sensitive to eigenvalue errors when the eigenvalues were derived from the other layer.

An alternative Fourier-technique-based junction temperature estimation method was reported in [39], where the time-domain power loss of the IGBT module was represented by a set of Fourier series, and the thermal impedance of the IGBT module was characterized in the frequency domain from the frequency response of a classical transient TEC. Then, the junction temperature of the IGBT was estimated in the frequency domain. However, that method still relied on the accuracy of the classical TEC.

#### 2.1.3 Numerical Models

Three-dimensional (3D) thermal models generated by commercial software, such as ANSYS, or numerical methods, such as convolution integrals over 3D geometry [40], can handle complex geometry and simulate realistic conditions of an IGBT module. They can provide accurate transient junction temperature estimation, if used with care. However, they are computationally intensive and are, therefore, impractical for online implementation.

#### 2.1.4 Digital Filter Models

As systems can be considered as filters, the thermal behavior of an IGBT module can be characterized by digital filters. In [41], a digital filter was obtained by discretizing a Foster-type TEC for efficient junction temperature estimation. However, that digital filter model still relied on the accuracy of the Foster-type TEC; and the continuousdiscrete conversion in that model might not preserve the properties of the continuoustime model at high frequencies (above 1/10 of the sampling frequency).

In [42], digital filters for modeling thermal cross-couplings in power devices were obtained by fitting the corresponding frequency-domain thermal characteristics, which were measured by applying pseudo-random binary sequence (PRBS) input-power waveforms to the power devices. However, the control of the generation of the PRBS power loss in IGBTs can be very difficult, especially when the thermal characteristics over a wide range of frequencies are of interest.

The thermal models for IGBT modules in the literature are compared in terms of accuracy, complexity, and computational cost, as shown in Table 2.2. It reveals that none of these models can have high accuracy while keeping the complexity and computational cost low. The TEC and digital filter models are the best options for real-time use.

Models	Accuracy	Complexity	Computational Cost
TEC	Low	Low	Low
Analytic model	Medium	High	Medium
Numerical model	High	Medium	High
Digital filter model	Low/medium	Low	Low

Table 2.2: Comparison of existing thermal models for IGBT modules

However, their accuracy needs to be improved.

### **2.2 Condition Monitoring Techniques for IGBT Modules**

The weak points in an IGBT module, in terms of aging, include the bond wires, emitter metallization, chip solder, and substrate solder [18], [43]. In the literature, the major aging processes of IGBT modules have been commonly monitored using the electrical and thermal parameters of the IGBT modules [44].

#### 2.2.1 Bond Wire Aging Monitoring Using V<sub>CEon</sub>

The wire bonding failures in IGBT modules are generally caused by bond wire liftoff, heel cracking, or metallization reconstruction.  $V_{CEon}$  has often been used for condition monitoring of bond wires and emitter metallization. As  $V_{CEon}$  is sensitive to  $T_j$ ,  $T_j$  should also be acquired. Detailed methods for  $T_j$  measurements will be presented in the next subsection. This subsection focuses on the  $V_{CEon}$  measurement circuitry.

The authors in [45] developed in situ  $V_{CEon}$  measurement circuitry for IGBTs on electric vehicles, as shown in Fig. 2.2. An auxiliary power supply unit (PSU) was used to inject 100 mA current for junction temperature measurements, and a high-current (50A) pulse train for diagnosis of a selected IGBT during a vehicle stop. The selection was achieved by using a selector relay network to connect the IGBT switch terminals to the PSU. The  $V_{CEon}$  measurement, however, cannot be performed when the inverter is operating. Extreme transient voltages and common-mode voltages that affect measurement accuracy and may cause damage to the sensitive measuring equipment have to be handled carefully.



Fig. 2.2: Schematic diagram of the in situ measurement circuitry for IGBTs on electric vehicles in [45].

In [46], an "on-line"  $V_{CEon}$  measurement scheme was invented, as shown in Fig. 2.3. Different auxiliary MOSFETs had to support the high DC voltage but conduct very low currents (100 mA or less) with respect to the power IGBT.  $V_{CEon}$  at 100 mA was acquired during the bridge "dead time" to indirectly measure IGBT junction temperature. Then, within a very short time (around 1 ms),  $V_{CEon}$  was measured at about 100 A. This circuit may acquire  $V_{CEon}$  at high current and at a desired junction temperature. However, the timing or synchronization of a sequence of  $V_{CEon}$  measurements is difficult to control.

In summary, the method of using  $V_{CEon}$  usually requires complex and costly circuitry to detect millivolt-level drifts in  $V_{CEon}$  against a noisy collector-emitter voltage  $V_{CE}$ , which varies from several volts to hundreds of volts during the operation of the IGBT. The implementation may interrupt the normal operation of power converter or alter the original power converter circuit design.



Fig. 2.3: *V<sub>CEon</sub>* measurement scheme (parallel diodes are not drawn) in [46].

# 2.2.2 Solder Aging Monitoring Using Temperature-Sensitive Electrical Parameters and *R*<sub>thjc</sub>

 $R_{thjc}$  has often been used for condition monitoring of solder layers [47], [48]. The detection of solder aging requires measurements of a temperature-sensitive electrical parameter (TSEP), e.g., the saturation collector-emitter voltage  $V_{CEon}$ , to indirectly measure  $T_j$ , which is then used to calculate the changes in  $R_{thjc}$  using (1.2).

In [49], common TSEPs were compared in terms of sensitivity, linearity, calibration need, genericity, accuracy, measurement of a thermal impedance, and on-line measurement. As shown in Fig. 2.4, voltage under a low current (e.g.,  $V_{CEon}$  at 100 mA) is the best TSEP overall. However, as discussed in the previous subsection, measuring  $V_{CEon}$  requires additional high-accuracy measurement circuitry and cannot be performed during the operation of the power converter.



Fig. 2.4: Comparison of the different TSEPs [49]. (a) Voltage under a low current (diode, IGBT, MOSFET in OFF-state). (b) Threshold voltage. (c) Voltage under a high current. (d) Gate-emitter voltage. (e) Saturation current. (f) Switching times.

# 2.2.3 Other Electric Aging Precursor and Aging Monitoring Techniques

Other aging monitoring methods based on monitoring changes in the switching behaviors of IGBTs, as a consequence of junction temperature elevation, have been reported in [50]-[52]. In [50], the dynamic changes in the gate current during the turn-on and turn-off processes of an IGBT module were used to monitor the aging of the module. In [51], a small change in the inverter harmonic current was used to detect the solder aging of the IGBT modules. In [52], the turn-off time was used as a precursor for IGBT

latch-up. However, in practice, the changes in these switching behavior-related parameters are very difficult to catch, especially when the increase in the junction temperature is not significant.

#### 2.2.4 Aging Monitoring Using Thermal Signal

Compared to electrical parameters, the changes in the temperatures of power converters are much slower and, therefore, can be more easily captured online. As a temperature signal consists of relatively low frequency components, the measurement noises can be easily filtered out. However, except for  $T_j$ , which is difficult to measure directly, other temperatures have rarely been used for monitoring the aging of IGBT modules.

In [53], a lookup table between power loss and case temperature was first constructed for a healthy IGBT module. Then, the power loss during the aging process of the IGBT module was calculated using the case-above-ambient temperature measured at one point and a one-dimensional (1D) case-to-ambient TEC model. The difference between the case temperature corresponding to the calculated power loss obtained from the lookup table and the measured case temperature was then used to estimate the increase in  $R_{thjc}$  for solder aging monitoring. That approach is based on the assumption that solder aging causes a rise of  $R_{thjc}$  and  $T_j$ , which causes an increase of the power loss. However, the increase of the power loss can be caused by either solder aging or bondwire aging and the power loss cannot be calculated accurately by the 1D case-temperature-based method when a substrate solder crack occurs.

The condition monitoring techniques for IGBT modules in the literature are compared in terms of performance, requirement for additional circuit, and feasibility for online implementation, as shown in Table 2.3. Using  $V_{CEon}$  for monitoring bond wire aging and measuring  $T_j$  and  $R_{thjc}$  are the only methods that can provide satisfactory performance. However, additional complex circuity is required and the operation of the power converter may be interrupted. The method that uses  $T_c$  for solder aging monitoring is the only method that is easy to implement. However, that method is limited to monitoring a simple device aging process—solder aging only.

#### 2.3 Summary

In this chapter, previous works on thermal models and condition monitoring techniques of IGBT modules were reviewed. These thermal models were discussed in terms of accuracy, complexity, and computational cost. A simple cost-effective thermal model suitable for online transient junction temperature estimation is still lacking. Condition monitoring for IGBTs is an emerging topic. The existing aging precursors and the corresponding measurement techniques are still impractical for real-world applications. Typical drawbacks are additional complex circuits, alternation of the existing power converter design, and interruption of the normal operation of the power converters.

Techniques	Performance	Additional complex circuit	Online implementation
Bond wire aging monitoring using $V_{CEon}$	Good	Yes	Difficult
Solder aging monitoring using TSEPs and <i>R</i> <sub>thjc</sub>	Poor for most TSEPs	Yes	Difficult for most TSEPs
IGBT switching behaviors	Poor	Yes	Difficult
Solder aging monitoring using one point $T_c$	Susceptible to interference	No	Easy

Table 2.3: Comparison of existing condition monitoring techniques for IGBT modules

# Chapter 3. A Physics-Based, Improved Thermal Equivalent Circuit Model

This chapter presents a physics-based, Cauer-type thermal equivalent circuit (TEC) with improved transient performance for IGBT modules [54]. A new concept, lumped capacitance approximation error, is introduced. Based on this concept, a method is proposed to determine the appropriate number of sublayers that should be subdivided from a layer in an IGBT module. For the bulky baseplate layer, an analytical expression of its thermal impedance is derived and simplified to a first-order transfer function, which can be represented by a thermal resistance and capacitance pair in the TEC. The proposed Cauer-type TEC model greatly improves the accuracy of the conventional Cauer-type TEC model for the transient junction temperature estimation of IGBT modules with a slightly increased order only.

## **3.1** The Lumped Parameters of the Cauer-Type TEC

Based on the lumped capacitance model, the distributed Cauer-type TEC is approximated by a low-order, lumped Cauer-type TEC. The conventional lumped Cauertype TEC is usually constructed based on the experience of the designer. Typically, the temperature difference across each layer in the heat flow direction in the IGBT module is neglected. Then the volume that the heat flows through in each layer is treated as a lump with uniformly distributed temperature. For a layer, its thermal resistance,  $R_{th}$ , and thermal capacitance,  $C_{th}$ , are calculated by

$$R_{th} = \int_0^{d_l} \frac{1}{k \cdot A(z)} dz \tag{3.1}$$

$$C_{th} = \int_0^{d_l} c_v \cdot A(z) dz \tag{3.2}$$

where  $d_l$  is the layer's thickness, *k* the thermal conductivity of the layer's material,  $c_v$  is the volumetric heat capacity of the layer's material, *z* is the variable representing the vertical distance from the top surface of the layer to the point of interest, and A(z) is the horizontal cross-section area at the distance *z*. A heat spreading angle (with a typical value of 45°) can be incorporated into (3.1) and (3.2) by adjusting A(z). For a "standard" IGBT module, shown in Fig. 3.1, thermal behavior can be modeled by a 7<sup>th</sup> order lumped Cauer-type TEC, as illustrated in Fig. 3.2. The conventional lumped Cauer-type TEC is computationally efficient. However, it may not be able to accurately capture the junction temperatures of IGBT modules in real-world applications, e.g., the estimation of the junction temperature variation of a power converter in a line cycle.



Fig. 3.1: The architecture of an IGBT module with Cauer-TEC RC parameters.



Fig. 3.2: The conventional lumped Cauer-type TEC for an IGBT module.

# **3.2** Construction of the Improved Cauer-Type TEC

The lumped Cauer-type TEC is built upon the lumped capacitance model. Thus, the estimated temperature distribution is discrete. As illustrated in Fig. 3.3, the steadystate temperature distribution in the layers of an IGBT module in the vertical direction zestimated by the conventional lumped Cauer-type TEC, i.e., the dash-line curve, is compared to the actual temperature distribution (i.e., the solid-line curve), where the top surface of the chip layer is the reference position with z = 0 and the horizontal temperature axis starts from  $T_c$  to the  $T_j$ . The actual temperature distribution through layers is continuous, as illustrated in Fig. 3.3, and can be estimated using a distributed Cauer-type TEC or an FEA model, which is more accurate than the lumped Cauer-type TEC. The slope of the temperature distribution in each thin layer is almost constant because the difference between the heat condition areas on the top and bottom surfaces of the layer, caused by the heat spreading effect, is negligible. The errors in the temperature distribution approximation using the conventional lumped Cauer-type TEC, i.e., the triangle shadow areas enclosed by the two curves, result in extra heat stored in the layers of the TEC, leading to an overestimation of the thermal capacitance of each layer. This results in an underestimation of the junction temperature changes in the transient states.



Fig. 3.3: Illustration of the temperature distributions in the layers of an IGBT module estimated by the conventional lumped Cauer-type TEC and the actual continuous temperature distribution of the IGBT module in the steady-state condition.

#### 3.2.1 Subdivision of the Layers in IGBT Modules

The performance of the lumped Cauer-type TEC can be improved by subdividing a bulky layer into smaller sublayers with the same thickness and modeling each sublayer by a smaller lump, which can reduce the temperature approximation error and the extra heat storage capacity. To determine whether a layer needs subdivision, a lumped– capacitance approximation error  $CE_{lc}$  is defined for the layer and compared with the total thermal capacitance  $C_{th,tot}$  of the IGBT module. Here  $CE_{lc}$  represents the overestimation error of the lumped thermal capacitance of the layer caused by the temperature distribution approximation error in the steady-state condition of the IGBT module, while  $C_{th,tot}$  is the amount of heat needed to raise  $(T_j - T_c)$  by 1 °C. Consider the *i*th layer. The corresponding  $CE_{lc,i}$  and  $C_{th,tot}$  are calculated by

$$CE_{lc,i} = \frac{1}{2N_{sub}} \frac{R_{th,i}}{R_{thjc}} C_{th,i}$$
(3.3)

$$C_{th,tot} = \frac{1}{R_{thjc}} \left[ \sum_{i=1}^{M} C_{th,i} \cdot \left( \sum_{j=1}^{i} R_{th,j} - \frac{1}{2} R_{th,i} \right) \right]$$
(3.4)

where  $N_{sub}$  is the number of sublayers of the *i*th layer, *M* is the number of layers of the IGBT module,  $R_{th,i}$  is the thermal resistance of the *i*th layer, and  $C_{th,i}$  is the lumped thermal capacitance of the *i*th layer. The derivation of (3.3) and (3.4) is provided in the Appendix. The thermal resistance and capacitance of each sublayer is about  $1/N_{sub}$  of the thermal resistance and capacitance of the *i*th layer, respectively. According to (3.3), the value of  $CE_{lc,i}$  is reduced by subdividing the layer and is inversely proportional to the number of sublayers  $N_{sub}$ . By specifying a desired value for  $CE_{lc,i}$ ,  $N_{sub}$  can be determined from (3.3) for each layer. For example, a criterion for building an accurate Cauer-type TEC is that  $CE_{lc}$  of each layer is less than 5‰ of  $C_{th,tot}$ .

The typical layer thicknesses in a "standard" IGBT module [17] and the corresponding  $R_{th}$ ,  $C_{th}$ , and  $CE_{lc}$  per chip are shown in Table 3.1, assuming a 10 mm × 10 mm × 0.2 mm chip size and a 45° heat spreading angle. According to the proposed criterion, it is straightforward to subdivide the aluminum nitride (AlN) layer into four sublayers. Moreover, since the transient temperature in milliseconds is largely determined by the thermal behavior of the silicon chip, it is suggested that the chip layer be subdivided into two to three sublayers to improve the fast transient performance of the

	$d_l$ (mm)	$R_{th}$ (°C/W)	$C_{th}$ (J/°C)	$CE_{lc}$ (J/°C)	$CE_{lc}/C_{th,tot}$ (%)
Chip	0.2	0.0161	0.0326	0.0021	2.1
Solder	0.05	0.0078	0.0086	2.6×10-4	0.26
Copper	0.3	0.0072	0.1113	0.0032	2.4
AlN	0.635	0.0275	0.1968	0.021	16.5
Copper	0.3	0.0052	0.1549	0.0032	2.4
Solder	0.2	0.0194	0.0554	0.0043	3.3
Baseplate	5	0.0438	5.4519	0.94	725.4

Table 3.1: Typical layer thicknesses and thermal parameters (without subdivision) in a high-power IGBT module.

Cauer-type TEC model.

#### **3.2.2** Approximation of the Thermal Impedance of the Bulky Baseplate

However, the subdivision of the bulky baseplate layer using the proposed criterion will dramatically increase the order of the TEC. To solve this problem, a physics-based analysis is presented to derive a low-order approximation of the thermal behavior of the baseplate. First, an exact description of the thermal behavior of the baseplate is derived by infinitely slicing the baseplate vertically, as shown in Fig. 3.4. This model could be described by the following equations [55]

$$\begin{cases} -\frac{dq(z_{bp})}{dz_{bp}} = sC_{th}(z_{bp})T(z_{bp}) \\ -\frac{dT(z_{bp})}{dz_{bp}} = R_{th}(z_{bp})q(z_{bp}) \end{cases}$$
(3.5)

where  $z_{bp}$  is the variable representing the vertical distance from the top surface towards the bottom surface of the baseplate;  $q(z_{bp})$ ,  $C_{th}(z_{bp})$ ,  $R_{th}(z_{bp})$ , and  $T(z_{bp})$  denote the heat flow, distributed heat capacitance, thermal resistance, and temperature at  $z_{bp}$ , respectively. The  $R_{th}(z_{bp})$  and  $C_{th}(z_{bp})$  can be calculated by

$$R_{th}(z_{bp}) = \frac{dz_{bp}}{k_{bp}[a+2z_{bp}\cdot\tan(\theta)][b+2z_{bp}\cdot\tan(\theta)]}$$
(3.6)

$$C_{th}(z_{bp}) = c_{vbp}[a + 2z_{bp} \cdot \tan(\theta)][b + 2z_{bp} \cdot \tan(\theta)] \cdot dz_{bp}$$
(3.7)

where *a* and *b* are the side lengths of the heat flux area at the top surface of the baseplate,  $k_{bp}$  is the thermal conductivity of copper,  $c_{vbp}$  is the volumetric specific heat of copper, and  $\theta$  is the heat spreading angle. The  $q(z_{bp})$  in (3.5) can be eliminated by

$$-\frac{d^{2}T(z_{bp})}{dz_{bp}^{2}} = \frac{dR_{th}(z_{bp})}{dz_{bp}}q(z_{bp}) + R_{th}(z_{bp})\frac{dq(z_{bp})}{dz_{bp}}$$
$$= \frac{dR_{th}(z_{bp})}{dz_{bp}}\left[-\frac{1}{R_{th}(z_{bp})}\frac{dT(z_{bp})}{dz_{bp}}\right] + R_{th}(z_{bp})\left[-sC(z_{bp})T(z_{bp})\right]$$
(3.8)

Then, by substituting (3.6) and (3.7) into (3.8) and some manipulations, the governing equation for  $T(z_{bp})$  is obtained by

$$\frac{d^2 T(z_{bp})}{dz_{bp}^2} + \frac{8\tan^2(\theta) \cdot z_{bp} + 2\tan(\theta) \cdot (a+b)}{[a+2z_{bp} \cdot \tan(\theta)][b+2z_{bp} \cdot \tan(\theta)]} \frac{dT(z_{bp})}{dz_{bp}} - \frac{sc_{vbp}}{k_{vbp}} T(z_{bp}) = 0$$
(3.9)

An explicit solution of (3.9) only exists for a = b. Since a and b in (3.9) are usually close, they can be replaced by  $c = \sqrt{ab}$ . Then, the general solution of (3.9) is given by

$$T(z_{bp}) = \frac{1}{1+2\tan(\theta) \cdot z_{bp}/c} \cdot \left[C_1 \cosh\left(\sqrt{sc_{vbp}/k_{bp}} \cdot z_{bp}\right) + C_2 \sinh\left(\sqrt{sc_{vbp}/k_{bp}} \cdot z_{bp}\right)\right]$$
(3.10)



Fig. 3.4: The exact description of the thermal behavior of the baseplate. (a) The sketch of the geometry and heat spreading and (b) the distributed Cauer-type TEC model for the baseplate.

where  $C_1$  and  $C_2$  are constants determined by the boundary conditions of the baseplate. As the time constant of the cooling system is much larger than that of the IGBT module, the temperature of the bottom surface of the baseplate can be assumed to be a constant. Then, the boundary conditions can be simplified to

$$T(d_{bp})=0$$

$$q(0)=-k_{bp}c^{2}\cdot\frac{dT(z_{bp})}{dz_{bp}}\Big|_{z_{bp}=0}=P_{flux}$$
(3.11)

where  $d_{bp}$  is the thickness of the baseplate and  $P_{flux}$  is heat flux applied at the top surface of the baseplate. Then, the transient thermal impedance  $Z_{th,bp}$  of the baseplate is

$$Z_{th,bp} = \frac{T(0) - T(d_{bp})}{P_{flux}} = \frac{1/(k_{bp}c^2)}{2\tan(\theta)/c + \sqrt{sc_{vbp}/k_{bp}} \cdot \coth(\sqrt{sc_{vbp}/k_{bp}} \cdot d_{bp})}$$
(3.12)

where *s* is the Laplace variable. Equation (3.12) can be further simplified by substituting the Taylor series of the term  $\operatorname{coth}(\sqrt{sc_{vbp}/k_{bp}} \cdot d_{bp})$ . For the baseplate layer, only the relatively low frequency components (typically 0 ~ 20 Hz) need to be considered. Then, the value of  $\sqrt{sc_{vbp}/k_{bp}} \cdot d_{dp}$  is small enough to allow the term  $\operatorname{coth}(\sqrt{sc_{vbp}/k_{bp}} \cdot d_{dp})$ approximated by the first two terms of its Taylor series. Then, (3.12) is reduced to

$$Z_{th,bp} = \frac{1/(k_{bp}c^2)}{1/d_{bp}+2\tan(\theta)/c+\frac{1}{3}sc_{vbp}d_{bp}/k}$$
  
=  $\frac{d/(k_{bp}c^2)}{1+2\tan(\theta)d_{bp}/c} \cdot \frac{1}{1+\frac{1}{3}d_{bp}/[k_{bp}\cdot c^2 \cdot (1+2\tan(\theta)d_{bp}/c)] \cdot sc_{vbp}d_{bp}c^2}$   
 $\approx R_{th,bp} \cdot \frac{1}{1+sR_{th,bp}C_{th,bp}/3}$  (3.13)

where  $R_{th,bp}$  and  $C_{th,bp}$  are the thermal resistance and the lumped thermal capacitance of the baseplate, respectively. Equation (3.13) suggests that the equivalent thermal capacitance for the baseplate approximates one third of  $C_{th,bp}$ . Therefore, the transient thermal behavior of the bulky baseplate can be modeled more accurately by still using a first-order transfer function, i.e., a single thermal resistance and capacitance pair.

#### **3.3** Simulation Validation

A commercial IGBT module, CM400DY-12NF, made by Powerex was studied to compare the transient performance of the conventional Cauer-type TEC and the proposed Cauer-type TEC. The IGBT module has two switches. Each switch consists of two IGBT chips and two diode chips.

A high-fidelity FEA thermal model of the test IGBT module cooled by a cold plate (see Fig. 3.5) was built as a reference model in Autodesk Simulation Multiphysics 2012, which is a commercial FEA software platform. The geometry information of the module was provided by Powerex. The IGBT module was mounted on a liquid-cooled cold plate, which is represented by an aluminum block with coolant pipes. The temperature of the pipes was set to the inlet coolant temperature. The thickness of the thermal grease layer between the baseplate of the module and the cold plate was 0.1 mm. In the FEA thermal model, a fine mesh was used for the IGBT module, while a relatively



Fig. 3.5: The model of a CM400-12NF IGBT module with a cold plate built in Autodesk Simulation Multiphysics.

coarse mesh with 20,044 nodes was used for the cold plate. The total mesh of the FEA thermal model of the IGBT module with the cold plate has 79,787 nodes. The power losses were assumed to be generated inside the chips close to their top surfaces [35]. The order of the conventional Cauer-type TEC is 7; while the order of the proposed Cauer-type TEC is 12, where the chip and AlN layers were subdivided into three and four sublayers, respectively.

First, the transient thermal impedances from junction to case of the IGBT module,  $Z_{thjc}(t)$ , were obtained from the conventional Cauer-type TEC model, the proposed Cauertype TEC model, and the FEA thermal model by applying a step-change power loss to the three models while setting the temperature of the bottom surface of the IGBT module to be a constant. The results are compared in Fig. 3.6. It shows that the proposed Cauertype TEC model was more accurate than the conventional Cauer-type TEC model with



Fig. 3.6: Comparison of the transient thermal impedances from junction to case obtained from the conventional Cauer-type TEC model, the proposed Cauer-type TEC model, and the FEA model.

respect to the FEA thermal model. In particular, the value of  $Z_{thjc}(t)$ , from approximately 0.1 s to 0.9 s, was mainly determined by  $Z_{th,bp}$ , i.e., the thermal behavior of the baseplate. Due to the use of a more accurate model (3.13) to characterize the thermal behavior of the baseplate, the value of  $Z_{thjc}(t)$  obtained from the proposed Cauer-type TEC model was a clear improvement over that obtained from the conventional Cauer-type TEC model, from approximately 0.1 s to 0.9 s.

The accuracy of the proposed Cauer-type TEC for the junction temperature estimation was evaluated by placing the IGBT module as a leg in a three-phase, two-level inverter. The operating condition of the inverter was the following: 400 V DC-link voltage, 400 A output current, a power factor of 0.8, a unity modulation index, 5 kHz switching frequency, and 60 Hz line frequency. Fig. 3.7(s) compares the values of the junction-to-case temperature difference,  $\Delta T_{jc}(t)$ , estimated by using the conventional Cauer-type TEC model, the proposed Cauer-type TEC model, and the FEA thermal model. Fig. 3.7(b) shows the temperature estimation errors of the conventional Cauer-type TEC model and the proposed Cauer-type TEC model with respect to the FEA thermal model. The maximum error of the proposed Cauer-type TEC model was 2.8 °C, which increased to 5.1 °C when using the conventional Cauer-type TEC model. The average error of the proposed Cauer-type TEC model was only about 30% of that of the conventional Cauer-type TEC model over the operating period of the inverter.



Fig. 3.7: Comparison of junction temperature estimation by the three models when the inverter was operated at a constant operating condition: (a)  $\Delta T_{thjc}(t)$ ; and (b) errors with respect to the FEA thermal model.

# 3.4 Summary

This chapter proposed a method for constructing an improved low-order Cauertype TEC model for IGBT modules. First, the lumped capacitance approximation error was introduced to determine the appropriate number of lumps needed to model a layer in an IGBT module. For the bulky baseplate layer that requires many lumps to model, an analytical expression of its thermal impedance was derived and simplified to a first-order transfer function, which was then represented by a thermal resistance and capacitance pair in the TEC. The proposed Cauer-type TEC model was validated by simulation studies for a commercial IGBT module. Compared to the conventional Cauer-type TEC, the proposed Cauer-type TEC has a slightly higher order but greatly improved the accuracy of the transient junction temperature estimation for IGBT modules.

# Chapter 4. Real-Time Junction Temperature Estimation Using Low-Order Digital Filters

This chapter presents a digital filter model to characterize the thermal behaviors of IGBT modules as well as heat sinks [57]. The thermal impedances from junction to ambient/coolant are characterized by low-order infinite impulse response (IIR) digital filters. Compared to the conventional Foster-type TEC model, the low-order IIR filter model can greatly improve the accuracy of the transient junction temperature estimation for IGBT modules. The IIR filters can be implemented easily and efficiently in digital processing systems for real-time junction temperature estimation of IGBT modules.

# 4.1 An IGBT Module with Liquid Cooling

An IGBT module typically contains multiple IGBT chips and diode chips, which are placed on one or several multilayer substrate stack(s) on a baseplate. The IGBT module with liquid cooling is illustrated in Fig. 4.1. During the operation of the IGBT



Fig. 4.1: An IGBT module (only one chip is shown) cooled by a cold plate.

module, heat is generated at the chips and spreads through the layers to the bottom of the baseplate, which is cooled by a cold plate. The thermal behavior of the IGBT module can be characterized by a linear system when the cooling condition is stable and the temperature effects of the materials' thermal properties are negligible.

#### 4.2 **IIR Digital Filter**

This section provides a brief introduction of the IIR filter. The use of digital filters instead of analog filters has many advantages, e.g., it is easy to design and implement digital filters, digital filters can handle larger dynamic ranges, and the performance of digital filters is stable [41]. Digital filters can be classified into two categories: finite impulse response (FIR) filters and IIR filters. The primary advantage of IIR filters over FIR filters is that they typically meet a given set of specifications with a much lower filter order [58]. The function of an IIR filter, H(z), in the z domain is

$$H(z) = \frac{B_{IIR}(z)}{A_{IIR}(z)} = \frac{b_0 + b_1 z^{-1} + \dots + b_{nb} z^{-nb}}{a_0 + a_1 z^{-1} + \dots + a_{na} z^{-na}}$$
(4.1)

where  $a_i$  (i = 0, 1, ..., na and  $a_0 \neq 0$ ) and  $b_j$  (j = 0, 1, ..., nb) are coefficients. The coefficients of H(z) for a target system can be easily determined by using the input u(t) and the output y(t) of the target system obtained in a certain system response and a filter parameter identification algorithm, such as the Steiglitz-McBride method. In this dissertation, the input and output during the impulse response of the target system are obtained from the step response of the system to identify the coefficients.

The Steiglitz-McBride method is a fast iterative algorithm that solves for all  $a_i$ and  $b_i$  simultaneously to minimize the total square error between the filter output and the given output signal. For example, the coefficients can be determined iteratively by solving the following least-square problem.

$$\min \sum_{t=1}^{N} \left\{ \hat{A}_{IIR,k+1}(z) \left[ \frac{1}{\hat{A}_{IIR,k}(z)} y(t) \right] - \hat{B}_{IIR,k+1}(z) \left[ \frac{1}{\hat{A}_{IIR,k}(z)} u(t) \right] \right\}^{2}$$
(4.2)

where k is the iteration index and N is the number of data.  $\hat{A}_{IIR,k}(z)$  and  $\hat{B}_{IIR,k}(z)$  are the estimates of the denominator and numerator in (4.1) obtained at iteration k;  $\hat{A}_{IIR,k+1}(z)$  and  $\hat{B}_{IIR,k+1}(z)$  are the refined estimates of denominator and numerator in (4.1) obtained by solving (4.2). The coefficients can also be identified when noises are present in input and output samples. More details of this algorithm can be found in [59], [60].

### 4.3 IGBT Thermal Modeling Using IIR Filters

In an IGBT module, the junction temperature of a chip depends on the heat generated by the chip (self-heating) and the heat generated by other chips (cross-heating). The self-heating of chips and the cross-heating between chips can be characterized by self-heating and cross-heating thermal impedances, respectively. The self-heating thermal impedances are typically provided in the datasheets of IGBT modules. The cross-heating thermal impedances can be obtained from FEA simulations and experiments. The thermal impedances can be considered as the step responses of the systems characterizing the self-heating of chips and cross-heating between chips [56]. Then, the impulse responses of the systems needed to design the IIR filters can be obtained from the derivatives of the transient thermal impedance curves. As the IIR filter uses the junction temperature as a feedback, accurate junction temperature estimation can be achieved by an IIR model with several orders. It is recommended that the order selected be the same as that of the Foster-type TEC model (typically 4-8) to ensure a much improved performance.

Given the impulse responses derived from the self-heating thermal impedances, the coefficients of the IIR filters characterizing the self-heating of chips can be easily obtained using the method mentioned in the previous subsection. Let  $H_{nn}(z)$  be the IIR filter charactering the self-heating of the *n*th chip. The implementation of the IIR filter in the *z* domain and the discrete time domain are expressed by (4.3) and (4.4), respectively.

$$T_{jn}(z) = H_{nn}(z)P_n(z) + T_a(z)$$
(4.3)

$$T_{jn}[k_{dt}] = \frac{1}{a_0} \left\{ \sum_{j=0}^{nb} b_j \cdot P_n[k_{dt} - j] - \sum_{i=1}^{na} a_i \cdot T_{jn}[k_{dt} - i] \right\} + T_a[k_{dt}]$$
(4.4)

where  $T_{jn}$  and  $P_n$  are the junction temperature and power loss of the *n*th chip, respectively;  $k_{dt}$  is the index of the discrete time sequence; and  $T_a$  is the ambient temperature.

The pure delay time due to the heat propagation path between chips needs to be considered to design the IIR filters to characterize the cross-heating between chips. This can be easily addressed by adding a discrete time delay term,  $z^{-c}$ , to H(z), where c is an integer, which is the division of the delay time and sampling period. Let  $H_{nnn}(z)$  be the IIR filter charactering the relationship between the junction temperature,  $T_{jn}$ , of the *n*th chip and the power loss,  $P_m$ , of the *m*th chip. Then, the implementations of the IIR filter in the z domain and the discrete time domain are expressed by (4.5) and (4.6), respectively.

$$T_{jn}(z) = z^{-c} H_{nm}(z) P_m(z) + T_a(z)$$
(4.5)

$$T_{jn}[k_{dt}] = \frac{1}{a_0} \left\{ \sum_{j=0}^{nb} b_j \cdot P_n[k_{dt} - j - c] - \sum_{i=1}^{na} a_i \cdot T_{jn}[k_{dt} - i] \right\} + T_a[k_{dt}] \quad (4.6)$$

When multiple chips are powered, their junction temperatures can be calculated by the superposition of the temperature rises from the ambient, caused by all of the chips. The z domain matrix form of the proposed method to calculate the junction temperatures of *N* chips of interest for an IGBT module with *M* heat sources is given as follows, where  $\mathbf{1}_{1\times N}$  is a *N*-dimensional column vector with all elements being 1.

$$\begin{bmatrix} T_{j1} \\ T_{j2} \\ \vdots \\ T_{jN} \end{bmatrix} = \begin{bmatrix} H_{11} & H_{12} & \dots & H_{1M} \\ H_{21} & H_{22} & \dots & H_{2M} \\ \vdots & \ddots & \vdots \\ H_{N1} & H_{N1} & \dots & H_{NM} \end{bmatrix} \begin{bmatrix} P_1 \\ P_2 \\ \vdots \\ P_M \end{bmatrix} + T_a \cdot \mathbf{1}_{1 \times N}$$
(4.7)

## 4.4 Simulation Validation

A commercial IGBT module CM400DY-12NF made by Powerex (described in Chapter 3) was studied to compare the transient performance of the Foster-type TEC model and the proposed IIR filter thermal model. The same high-fidelity FEA thermal model of the test IGBT module cooled by a cold plate built in Autodesk Simulation Multiphysics 2012 was used as the reference model. A detailed description of the FEA thermal model can be found in Chapter 3. In the study described here, the inlet coolant temperature was set to 60 °C.

As the cross-heating between chips on different stacks is negligible, only the thermal coupling effect between IGBT chips and diode chips on the same stack was considered. A step-change power loss was applied to the two IGBT chips in parallel on a stack to obtain the self-heating thermal impedance from the IGBT junction to ambient,  $Z_{thjaQQ}(t)$ . Another simulation, in which a step-change power loss was applied to the two diode chips on the same stack, was performed to obtain the cross-heating thermal impedance from the IGBT junction to ambient,  $Z_{thjaQQ}(t)$ . Two 5th-order digital IIR filters were designed to characterize the two thermal impedances, respectively. The sampling period for the implementation of the IIR filters was 1 ms. The coefficients of the digital IIR filters were obtained by using the built-in Steiglitz-McBride method, stmcb(), in

MATLAB<sup>®</sup>. Two 6th-order Foster-type TECs were also obtained by fitting the responses of the two thermal impedances using the curve fitting toolbox of MATLAB. The performance of the digital IIR filters and the Foster-type TECs were compared with respect to the FEA thermal model, as shown in Fig. 4.2. The proposed IIR filter thermal model agrees well with the FEA thermal model for characterizing both  $Z_{thjaQQ}(t)$  and  $Z_{thjaQD}(t)$ . However, noticeable errors were observed in  $Z_{thjaQQ}(t)$  characterized by the Foster-type TEC model when t < 0.07 s. The performance of the Foster-type TEC model for  $Z_{thjaQD}(t)$  characterization is even worse–erroneous when t < 0.4 s–because the Fostertype TEC model cannot model the pure delay caused by the distance between the IGBT and diode chips.

The accuracy of the proposed IIR filter thermal model for junction temperature estimation was further evaluated in a power converter operating scenario. The IGBT module was operated as a leg in a two-level three-phase inverter. The operating point was: 400 V DC-link voltage, 400 A output current, power factor of 0.8, unity modulation index, 5 kHz switching frequency, and 60 Hz line frequency. The  $T_j(t)$  responses estimated by the FEA thermal model, the proposed IIR filter thermal model, and the Foster TEC model are compared in Fig. 4.3. The maximum error between the FEA thermal model and the proposed IIR filter thermal model was less than 1.5 °C. The Foster-type TEC model can track the trend of a  $T_j(t)$  increase but fails to estimate the fast  $T_j(t)$  variation in each fundamental period of the inverter. This is because Foster-type TECs cannot accurately characterize the thermal impedances, especially  $Z_{dhjaQQ}(t)$  and  $Z_{dhjaQD}(t)$  when t is small.



Fig. 4.2: Comparison of the thermal impedances modeled by the 6th-order Foster-type TECs, the proposed 5th-order IIR filters, and the FEA method: (a)  $Z_{thjaQQ}(t)$  and (b) $Z_{thjaQD}(t)$ .



Fig. 4.3: Comparison *T<sub>j</sub>(t)* of the IGBT estimated by the three methods when the inverter is operated at a constant operating condition:(a) junction temperature and (b) error with respect to the FEA method.

A transient load, which mimicked the starting process of an electric motor in a hybrid electric vehicle (HEV), was applied to the inverter to evaluate the accuracy of the proposed IIR filter thermal model for real-time junction temperature estimation. The rotor speed of the motor was accelerated from 0 rpm at 0 s to 500 rpm at 1 s. The currents through the IGBT module during this period were generated by simulating an HEV powertrain using a battery model provided by the MATLAB/Simulink<sup>®</sup> example library. The power losses,  $P_{loss}(t)$ , generated by the chips were calculated based on the simulated currents as well as the conduction voltage-current curve and switching loss-current curve provided by the datasheet for the IGBT module. The total power loss profile of the IGBT chips in one switch of the module is shown in Fig. 4.4, where the power loss is averaged over each switching period.



Fig. 4.4: Total power loss profile of the IGBT chips in one switch of the IGBT module in a transient load condition.





Fig. 4.5: Comparison  $T_j(t)$  estimated by the three methods when the inverter is operated with a transient load: (a) junction temperature; and (b) error at peaks and valleys with respect to the FEA method.
The values of  $T_j(t)$  estimated by the FEA thermal model, the proposed IIR filter thermal model, and the Foster-type TEC model are compared in Fig. 4.5. The proposed IIR filter thermal model agrees with the FEA thermal model much better than the Fostertype TEC model. The maximum error of the IIR filter thermal model is only 1 °C when  $T_j(t)$  varies between 60 °C and 87 °C, while the maximum error of the Foster-type TEC model is about 5 °C. The proposed IIR thermal model can always provide an accurate  $T_j(t)$  estimation when both the amplitude and frequency of the  $P_{loss}(t)$  vary. In contrast, the Foster-type TEC model overestimates  $T_j(t)$  fluctuations when the frequency is low but underestimates  $T_j(t)$  fluctuations after the frequency increases, indicating an unstable performance during the transient load.

## 4.5 Summary

This chapter presented the use of low-order IIR digital filters for accurate realtime junction temperature estimation of IGBT modules. The self-heating and crossheating thermal impedances of the IGBT module were characterized by low-order IIR filters. The coefficients of the IIR filters were obtained by using the Steiglitz-McBride method. Simulation results have shown that the transient junction temperature of IGBT modules can be accurately estimated by using the IIR filters with orders even lower than those of the Foster-type TECs. The digital IIR filters can be easily implemented in a digital processing system for real-time junction temperature estimation of IGBT modules.

# Chapter 5. An Effective Heat Propagation Path-Based Online Adaptive Thermal Model

Information on the junction temperature is crucial for operational management of IGBT modules. In practice, junction temperature is typically estimated by using an electrothermal model. IGBT modules are subject to various aging processes during operation, some of which, e.g., substrate solder cracking, change the thermal impedance of an IGBT module. However, little work has been reported in the literature on including aging effects in online thermal behavior modeling of IGBT modules.

This chapter presents an effective heat propagation path (EHPP)-based online adaptive thermal model for IGBT modules [61], where the EHPP is proposed to quantify the impact of substrate solder cracks on heat propagation inside IGBT modules. A straightforward relationship between substrate solder crack and the degree of nonuniformity of case temperature distribution is established. Based on the EHPP, the parameters of a TEC, e.g., a Cauer-type TEC, are adjusted online to track thermal behavior changes of the IGBT modules caused by substrate solder cracks, leading to an adaptive thermal model. The proposed adaptive thermal model is validated by comparing its results with FEA simulation results for a commercial IGBT module.

## 5.1 EHPP-Based Thermal Behavior Characterization for IGBT Modules

Due to the complexity of the geometry and differences in the materials' properties, the heat propagation inside an IGBT module is a complex, dynamical process. In this section, a new concept of EHPP is introduced to effectively quantify the impact of substrate solder cracks on the thermal behavior of IGBT modules. Then, the dissertation will show that the change in the thermal resistance from the junction to the case of an IGBT module caused by substrate solder cracks can be interpreted by the change in the EHPP, which can be quantified by the degree of nonuniformity of the case temperature distribution.

The EHPP is defined as the thermal path in an IGBT module through which most heat flows, as illustrated in Fig. 5.1. During operation of an IGBT, heat is assumed to be generated at the top surface of a die and to spread down through different layers to the bottom of the baseplate, which is cooled by a heat sink or a cold plate. Compared to the top surface of a die, the EHPP covers a larger area at the bottom surface of the baseplate,



Fig. 5.1: The EHPP in a healthy IGBT module.

called the baseplate hot area,  $A_{hot}$ . The temperature inside  $A_{hot}$  is much higher than that in the remaining area of the bottom surface of the baseplate. The baseplate hot area,  $A_{hot}$ , can be viewed as the thermally effective contact to the heat sink or the effective heat removal area. The following paragraphs describe how the EHPP and  $A_{hot}$  change with substrate solder aging and how the relationship between the change in case temperature distribution and the remaining noncracking solder area in the substrate solder layer is derived.

A crack usually initializes from the edge of the substrate solder layer and then propagates to the center [71]-[73]. The crack does not significantly affect the heat propagation inside the module or the thermal resistance of the module until it encroaches on the EHPP. When that occurs, the status of the IGBT module is dangerous as the heat flows are forced to concentrate in the remaining noncracking portion of the substrate solder layer. Both the EHPP and  $A_{hot}$  shrink, resulting in an increase in the IGBT module's thermal resistance and the degradation of heat removal at the bottom surface of the baseplate, as illustrated in Fig. 5.2.

Assuming that the same amount of heat is generated at the top surface of the chip,



Fig. 5.2: The EHPP in an IGBT module when cracks developed from the edge to the center in the substrate solder layer.

the shrinkage of the  $A_{hot}$  can be indicated by an increase in the difference between the case temperature, i.e.,  $T_{cchip}$ , at a location right beneath the die (i.e., in the middle of the  $A_{hot}$ ) and the case temperature, i.e.,  $T_{cside}$ , at a location inside but near the edge of the  $A_{hot}$ . If the  $A_{hot}$  shrinks,  $T_{cchip}$  will increase due to greater concentration of the heat flows, while  $T_{cside}$  will drop as its location tends to move out of the  $A_{hot}$  or away from the main EHPP. Obviously, it is important to choose appropriate locations at the bottom surface of the baseplate to acquire the nonuniformity of the case temperature distribution related to a change in the EHPP.

However, establishing the relationship between the difference of  $T_{cchip}$  and  $T_{cside}$ ; the remaining noncracking area,  $A_r$ , in the substrate solder layer; and the change in the EHPP caused by substrate solder cracks is still not straightforward, as an IGBT module involves multilayer thermal conduction and heat storage and spreading. To establish an effective relationship, this chapter simplifies the problem to a baseplate layer inverse heat conduction problem (IHCP), which utilizes the measurable case temperatures to inversely estimate the remaining noncracking solder area. By neglecting the heat spreading in the thin substrate solder layer, the heat flux flowing through the substrate solder layer to the baseplate is perpendicular to the baseplate top surface. Then, the  $A_r$  in the substrate solder layer is equivalent to the heat flux area on the top surface of the baseplate.

Fig. 5.3 shows the change in the heat flux on the baseplate top surface due to the aging of the substrate solder. It is assumed that the cooling condition is stable such that the boundary condition on the bottom surface of the baseplate is stationary. Since the heat flux is concentrated at the center of the baseplate top surface and the total heat propagating through the EHPP at a certain operating condition is specific, the IHCP



Fig. 5.3: IHCP using case temperature to estimate the effective heat flux area.

solution,  $A_r$ , is unique. Thus, a look-up table can be conveniently built via a numerical method, such as FEA, to relate ( $T_{cchip}-T_{cside}$ ) to  $A_r$  for various operating conditions for online applications.

## 5.2 The EHPP-Based Online Adaptive Thermal Model for IGBT Modules

Once the health condition of the substrate solder of an IGBT module is quantified online using  $A_r$ , the next problem is how to model the thermal behavior of the IGBT module adaptively by taking into account the substrate solder aging effect. The FEA method can provide a high-fidelity thermal model which, however, requires extensive time and memory to implement. This dissertation proposes an EHPP-based online adaptive RC TEC model for IGBT modules. In the proposed model, a traditional RC TEC, e.g., a Cauer-type TEC, is used to physically represent each layer of the EHPP inside an IGBT module. The EHPP, shown as the area within the dashed lines in Fig. 5.4, is approximated by using the heat spreading angle technique [74]. The impact of substrate solder cracks on the thermal behavior of the IGBT module is interpreted via the change in the EHPP. Assuming that only the EHPP through the substrate solder layer and its two adjacent layers, i.e., the substrate copper layer and baseplate layer, is altered by the substrate solder cracks, then the area  $A_c$  of the EHPP through the bottom surface of the ceramic layer is constant.

The heat spreading angles,  $\theta$ , in degrees in the pattern layer and the ceramic layer, can be calculated by [75]:

$$\theta = 90 \tanh\left\{0.355 \left(\frac{\pi k}{180}\right)^{0.6}\right\}$$
(5.1)

where k is the thermal conductivity in W/(m °C) of the material. The heat spreading angle in the substrate solder layer is neglected owing to the relatively low thermal conductivity of solder and the relatively thin thickness of the substrate solder layer. The heat spreading



angle in the baseplate is cooling-condition dependent and is given as a function of the ratio between the total thickness of all of the heat-spreading layers beneath the chip and the side length of the chip [55]. Therefore, only the heat spreading angle,  $\alpha_{adj}$ , in the substrate copper layer needs to be determined according to the status of the substrate solder crack. Once the value of  $A_r$  in the substrate solder layer is obtained,  $\alpha_{adj}$  can be calculated using the geometry information,  $A_c$  and  $A_r$ , and the thickness,  $d_{pattern}$ , of the substrate copper layer.

The thermal resistance of each layer is calculated by:

$$R_{th,i} = \int_0^{d_i} \frac{1}{k_i \cdot A_i(z)} dz$$
 (5.2)

where  $R_{th,i}$ ,  $d_i$ ,  $k_i$ , and z are the thermal resistance, thickness, thermal conductivity of the material, and vertical distance from the top surface to the bottom surface of the *i*th layer, respectively; and  $A_i(z)$  is the horizontal cross-section area at a distance z, which is calculated according to the geometry information, including the heat spreading angle, of the layer. The calculation for the thermal capacitance of each layer, however, is not straightforward. Detailed information on the calculation of thermal impedances of electronic packaging can be found in [30], [76].

The approximated EHPP in the substrate solder layer and the adjacent copper layer and baseplate is adjusted according to  $A_r$ . Based on the approximated EHPP, the RC parameters of the three layers can be adjusted for the Cauer-type TEC, leading to an adaptive thermal model. The schematic of the proposed online adaptive TEC model is illustrated in Fig. 5.5, where  $I_c$ ,  $f_{sw}$ , PF,  $P_{loss}$ , and  $T_a$  are the collector current, switching frequency, power factor, power loss, and ambient temperature of the IGBT module, respectively.



Fig. 5.5: Schematic of the proposed online adaptive TEC model.

## **5.3 Simulation Validation**

A commercial IGBT module, CM1400DU-24NF, made by Powerex was studied to validate the proposed EHPP-based adaptive RC TEC model. The geometry information of the module was provided by Powerex. The IGBT module was in a halfbridge configuration. Each IGBT switch consisted of nine parallel dies placed on three DBC transistor stacks. Taking advantage of the symmetry of the stack layout, one-sixth of the module (41.5 mm  $\times$  75 mm), i.e., a thermal stack with three IGBT dies, was built in ADINA, a commercial FEA software platform, as shown in Fig. 5.6. The cold plate was simplified to an aluminum block with the bottom surface temperature set to be the inlet coolant temperature. The thermal interface between the baseplate and the cold plate was a thermal grease layer with a thickness of 0.1 mm.

The substrate solder was assumed to have a uniform crack from the edge to the center. Fig. 5.7 illustrates the geometry of the module viewed from the baseplate with the locations of the case temperature measuring points. Since the three dies were connected in parallel in one switch, the power losses in the three dies were assumed to be the same. The EHPP of the center die was assumed to be unchanged during the substrate solder cracking process. The approximated EHPPs of the three dies, as shown in Fig. 5.8, are separated by dashed lines. It should be noted that Fig. 5.8 is for illustration purposes only and does not reflect the real geometry of the module.



Fig. 5.6: The model of one-sixth of a CM1400 IGBT module built in ADINA.



Fig. 5.7: Baseplate side view of the IGBT module.

The effectiveness of using the nonuniformity of the case temperature distribution  $(T_{cchip}-T_{cside})$  as an indicator of substrate aging was evaluated for three solder crack cases at two specific operating conditions using the FEA method. The first operating condition was a constant load, i.e., constant power loss, case, where the test IGBT was conducted long enough to reach a steady-state temperature. The second operating condition was a dynamic load case, where the test IGBT was operated as a leg in a three-phase inverter with the collector current  $I_c = 450$  A, switching frequency  $f_{sw} = 2$  kHz, and power factor PF = 0.85. The AC output frequency of the inverter was 1 Hz, as an accurate thermal



Fig. 5.8: EHPPs of the side dies and the center die (separated by dashed lines).

modeling of the IGBT module is important as the IGBT undergoes more severe thermal stress under a low-frequency load [77]. The power loss profiles of one die for the two load conditions are shown in Fig. 5.9, where the power loss was averaged over each switching period in the dynamic load case. Fig. 5.10 shows the changes in nonuniformity of the case temperature distribution ( $T_{cchip}$ - $T_{cside}$ ) and the changes in maximum junction temperature,  $T_{jmax,side}$ , of the side die for different solder crack cases with respect to those of the healthy case. The case temperature was acquired after the maximum junction temperature was reached while considering the phase delay between the power loss and the measured case temperatures.

It can be seen that nonuniformity of the case temperature distribution and the junction temperature increased with the development of the substrate solder crack in both constant and dynamic load conditions. The increases in  $(T_{cchip}-T_{cside})$  and  $T_{jmax,side}$  in the 1.75-mm solder crack case were noticeable. However, in the other two cases where the



Fig. 5.9: Power loss profiles of constant load and dynamic load cases for one IGBT chip.



Fig. 5.10: Changes in nonuniformity of the case temperature distribution and maximum junction temperature in the three substrate solder crack cases in constant and dynamic load conditions.

solder cracks became more severe, both  $(T_{cchip}-T_{cside})$  and  $T_{jmax,side}$  increased more noticeably. These results proved that  $(T_{cchip}-T_{cside})$  is a sensitive indicator of the substrate solder aging process.

The thermal resistance  $R_{thja,side}$  of the side die was a key concern, since the solder crack would first encroach the EHPP of the side die. Using the proposed method, the value of  $R_{thja,side}$  was calculated. The thermal resistance of the thermal grease layer was also calculated by using (5.2). However, the heat spreading in the thermal grease layer was neglected; and, therefore, the cross-section area was equal to  $A_{hot}$ . The change in thermal resistance of the cold plate was not considered due to its negligible impact on the total thermal resistance. Fig. 5.11 compares the values of  $R_{thja,side}$  calculated by the FEA method and the proposed model. The results showed a good match between the two models, where the error between the two models was less than 2%. To examine the performance of the proposed model for adaptive dynamic junction temperature estimation, the results obtained from the FEA method, the proposed adaptive model, and the traditional Cauer-type TEC model without parameter adaption (i.e., using the fixed parameters extracted for the IGBT in the healthy condition) are compared in Fig. 5.12 under the dynamic load condition for the 3.5-mm crack and 5-mm crack cases. The relative error was defined as the ratio between the absolute error of the junction temperature estimated by the proposed model or the traditional Cauer-type TEC model with respect to the junction temperature obtained from the FEA method and the mean value of the junction temperature rise from coolant temperature estimated from the FEA method. The maximum error between the proposed model and the FEA model was less than 4 °C, and the relative errors were within 5% during most of the simulations in both cases. On the other hand, the maximum error between the traditional Cauer-type TEC model and the FEA model is more than 7°C, and the relative errors are much larger



Fig. 5.11: Comparison of  $R_{thja,side}$  calculated by the FEA method and the proposed model.



Fig. 5.12: Comparison of different models for the cases of 3.5-mm solder cracks and 5-mm solder cracks under the dynamic load condition: (a) junction temperature (3.5-mm solder crack); (b) junction temperature (5-mm solder crack); (c) relative error (3.5-mm solder crack); and (d) relative error (5-mm solder crack).

than those of the proposed model. These results clearly show that the proposed adaptive thermal model is much better than the traditional TEC models for junction temperature prediction as the proposed model includes the substrate solder aging effect by using the proposed EHPP method.

## 5.4 Summary

This chapter presented an EHPP-based RC TEC model for IGBT modules, where the parameters were adapted online to substrate solder cracks inside the IGBT modules. The concept of EHPP was introduced to effectively characterize the thermal behavior of IGBT modules using the heat spreading angle technique. The impact of substrate solder cracks on the thermal behavior of an IGBT module was interpreted by using the change in the EHPP as approximated based on the change in the nonuniformity of the case temperature distribution. The approximated EHPP was then used to adapt the RC parameters of the TEC online. Numerical simulation studies validated the effectiveness of using the nonuniformity of the case temperature distribution to indicate the severity of substrate solder cracks and the effectiveness of using the proposed adaptive thermal model to achieve improved junction temperature estimation with respect to traditional TEC models. The proposed model can be used for effective online condition monitoring, control, and thermal and health management of IGBTs and associated power electronic systems.

# Chapter 6. Real-Time Health Monitoring for IGBT Modules Using Case Temperatures

The major aging mechanisms in an IGBT module have two thermal consequences. First, some aging processes, e.g., bond wire liftoff and metallization degradation, lead to an increase in  $V_{CEon}$  and, therefore, an increase in power loss and heat generation in the IGBT module. Moreover, some aging processes, such as substrate solder crack, will deteriorate heat propagation inside an IGBT module, causing an elevation of  $T_j$ . Compared to electrical parameters, the changes in power converter temperature are much slower and, therefore, can be more easily captured online. As a temperature signal consists of relatively low frequency components, the measurement noises can be easily filtered out. However, except for  $T_j$ , which is difficult to measure directly, other temperatures have rarely been used for monitoring the aging of IGBT modules.

This dissertation proposes a new method of using two-dimensional (2D) case temperature for monitoring the aging of substrate solder, bond wires, and emitter metallization in IGBT modules in real time [78]. The proposed method is based on two facts: 1) substrate solder aging alters the heat flow inside an IGBT module and, therefore, the nonuniformity of the 2D case temperature distribution [61]; and 2) the bond wire and emitter metallization aging increases  $V_{CEon}$  and, therefore, the power loss of an IGBT module.

The proposed method first characterizes the nonuniformity of the 2D case temperature distribution in order to monitor substrate solder aging. Then, the proposed method utilizes linear system analysis of the thermal behavior of the IGBT module to determine the increase in heat generation, from which the increase in  $V_{CEon}$  is obtained to monitor bond wire and emitter metallization aging. In practical applications of the proposed method, different aging processes of an IGBT module can be monitored separately by placing two thermal sensors for each switch at the interface between the baseplate of the IGBT module and the cold plate.

To the authors' knowledge, this is the first work which uses 2D case temperatures to monitor online the major aging processes in IGBT modules. The proposed method was validated by simulation and experimental results for a commercial IGBT module.

## 6.1 Linear Thermal Behavior of IGBT Modules

Although the aging processes of IGBT modules are mainly driven by thermalrelated stresses and result in changes in the heat generation and temperature of IGBT modules, thermal parameters have rarely been used as precursors for monitoring the aging of IGBT modules. This is partly due to the difficulty in obtaining accurate information on some important thermal parameters, such as the  $T_j$ , during the operation and aging of IGBT modules. Moreover, some easy-to-measure temperatures, such as  $T_c$ , have not received enough attention in regard to monitoring the aging of IGBT modules. This might be due to the lack of a thorough study on the changes in thermal behavior caused by major aging mechanisms and the complexity in thermal modeling for IGBT modules. The linear thermal behavior of IGBT modules is discussed in this section. This discussion will serve as the basis for establishing the principle of the proposed method of using 2D case temperatures for real-time monitoring of the major aging processes of IGBT modules.

An IGBT module typically has a multilayered structure. The thermal behavior of an IGBT module is complex due to the complex geometry and differences in the materials' properties. The thermal behavior of an IGBT module in a certain health condition can be accurately modeled by a high-order, 3D, Cauer-type RC TEC based on the geometry and materials of the module. The RC parameters of the thermal model are assumed to be constant if the performance of the cooling system is stable and the IGBT works with the junction temperature in the allowed operating range in which variations of the materials' thermal properties, caused by junction temperature variations (called the temperature effect), are negligible. Under this circumstance, the thermal behavior of the IGBT module is linear during normal operation. Thus, the superposition principle can be used to model the temperature rise from the coolant to any point in the IGBT module caused by multiple sources of power loss in the IGBT module, such as the IGBT chips and the free-wheeling diode (FWD) chips. Since the power losses generated by the FWD chips are relatively small and their contribution to the case temperature rise as the IGBT module ages is negligible, only the power losses generated by IGBT chips are considered during the aging process. Then, as the thermal behavior of the IGBT module changes over time during the aging process, the temperature rise (from ambient to any point at the case) of the IGBT module has a linear time-varying relationship with the total power loss generated by the IGBT chips.

### 6.2 2D Case Temperature-Based Monitoring of IGBT Module Aging

The power loss,  $P_{loss}(T_j)$ , of a switch in an IGBT module includes the conduction loss,  $P_{Closs}(T_j)$ , and the switching loss,  $P_{SWloss}(T_j)$ , of the switch, which typically consists of multiple chips in parallel. The power losses depend on  $T_j$ , which is difficult to measure in practice. The values of  $T_j$  and  $P_{loss}(T_j)$  change with the aging of the IGBT module. The initial power loss,  $\hat{P}_{loss}(\hat{T}_j)$  (including  $\hat{P}_{Closs}(\hat{T}_j)$  and  $\hat{P}_{SWloss}(\hat{T}_j)$ ), and initial junction temperature,  $\hat{T}_j$ , are the values of the IGBT module when it is new (healthy); and they do not change with the aging of the IGBT module. The values of  $\hat{P}_{loss}(\hat{T}_j)$  and  $\hat{T}_j$  can be obtained by an iterative electrothermal modeling process using the datasheet parameters of the IGBT module, the power loss calculation equations in [62] or a power loss lookup table, and a thermal model for the IGBT module and the cold plate. When the IGBT module is healthy, the initial  $\hat{T}_j$  is the same as  $T_j$ , then

$$P_{loss}(T_j) = \hat{P}_{loss}(\hat{T}_j) \tag{6.1}$$

Unlike the junction temperature, the case temperature of an IGBT module can be directly measured and, therefore, is a good thermal parameter for monitoring the aging of IGBT modules in practical applications. First, consider a 1D RC TEC, in which the relationship between the steady-state 1D case temperature,  $T_c$ , and  $P_{loss}(T_j)$  is given by

$$T_c = P_{loss}(T_j) \cdot R_{thca} + T_a \tag{6.2}$$

where  $R_{thca}$  is the thermal resistance from case to ambient.

According to (6.2), a change in  $T_c$  reflects a change in the power loss and/or  $R_{thca}$ during the aging process. However,  $T_c$  in (6.2) represents the average case temperature, which is difficult to measure by using one or multiple thermal sensors in practice, especially when the nonuniformity of the case temperature distribution changes during the aging process of the substrate solder [61]. To solve the problem of using a 1D case temperature, this dissertation proposes to monitor the major aging processes of an IGBT module by using 2D case temperatures.

An IGBT module cooled by a cold plate with selected points for 2D case temperature measurements is illustrated in Fig. 6.1. Assuming that the IGBT module has linear thermal behavior, the steady-state temperature,  $T_{cp}$ , at any point p(x, y) on the bottom surface of the baseplate can be expressed by the following equation

$$T_{cp} - T_a = R_{eqcp} \cdot P_{loss}(T_j) \tag{6.3}$$

where  $R_{eqcp}$  is the equivalent thermal resistance characterizing the temperature rise from ambient to the point *p* when the IGBT module has a certain power loss,  $P_{loss}(T_j)$ . Equation (6.3) can be extended to the steady-state 2D case temperatures,  $\mathbf{T}_C$ , for all of the points of interest on the baseplate as follows:

$$\mathbf{T}_{C} - T_{a} \cdot \mathbf{J} = \mathbf{R}_{eqC} \cdot P_{loss}(T_{j})$$
(6.4)

where **J** is the matrix of ones, and  $\mathbf{R}_{eqC}$  is the matrix of the equivalent thermal resistances



• Selected points at the bottom surface of the baseplate for 2D case temperature measurements

Fig. 6.1: An IGBT module cooled by a cold plate with selected points for 2D case temperature measurements.

from the points of interest on the 2D bottom surface of the baseplate to ambient. The aging processes of an IGBT module result in changes in  $\mathbf{R}_{eqC}$  and/or  $P_{loss}(T_j)$ . This dissertation proposes a method of using the measured  $\mathbf{T}_C$  and  $T_a$  to monitor the changes in  $\mathbf{R}_{eqC}$  and  $P_{loss}(T_j)$  by (6.4) for real-time monitoring of the major aging processes of IGBT modules.

#### 6.2.1 Substrate Solder Aging

The fatigue of solder joints is a major failure mechanism of IGBT modules. Compared to the chip solder, the substrate solder is more critically subject to fatigue, commonly leading to solder cracks [63]. Substrate solder cracks shrink the area for heat conduction in the substrate layer, which deteriorates the heat propagation inside the IGBT module and increases the value of  $R_{thjc}$ . This will intensify the nonuniformity of the case temperature distribution at the bottom surface of the baseplate [61]. As a consequence,  $\mathbf{R}_{eqC}$  changes over time during the substrate solder aging process. The ratio ( $k_{hc}$ ) between an element  $R_{eqCh}$  of the matrix  $\mathbf{R}_{eqC}$  for a hot spot  $h(x_1, y_1)$  and an element  $R_{eqCc}$  of  $\mathbf{R}_{eqC}$ for a relatively cold point  $c(x_2, y_2)$  on the baseplate is defined as.

$$k_{hc} = \frac{R_{eqCh}}{R_{eqCc}} \tag{6.5}$$

The value of  $k_{hc}$  characterizes the impact of the alternation of the heat flow inside an IGBT module caused by substrate solder aging on the nonuniformity of the 2D case temperature distribution. As the substrate solder ages, the value of  $k_{hc}$  increases over time. Therefore,  $k_{hc}$  is a good parameter for real-time monitoring of the substrate solder aging process. An advantage of using  $k_{hc}$  is that it is independent of the operating condition of the IGBT module.

#### 6.2.2 Bond Wire and Emitter Metallization Aging

The aging of bond wires and metallization in an IGBT module results in an increase of  $V_{CEon}$  and, therefore, an increase in the total conduction loss of IGBT chips in a switch. A commonly used criterion indicating that the bond wire and emitter metallization are aged is that  $V_{CEon}$  has increased by 10~20% [64], which corresponds to a 10~20% increase in conduction loss. During this aging process,  $\mathbf{R}_{eqC}$  is constant, while  $P_{loss}(T_j)$  increases over time. By monitoring  $\mathbf{T}_C$  and  $T_a$ ,  $P_{loss}(T_j)$  can be estimated using (6.4). As the switching loss can be calculated using the equations in [62] with the information of  $T_j$ , the conduction loss can be obtained; and then  $V_{CEon}$  can be calculated [62] to examine the aging condition of the bond wires and emitter metallization.

#### 6.2.3 Mixed Aging Mode of Bond Wires, Metallization, and Substrate Solder

When an IGBT module undergoes multiple aging processes, both  $P_{loss}(T_j)$  and  $\mathbf{R}_{eqC}$  may change over time. Let  $\alpha_P$  be the ratio between  $P_{loss}(T_j)$  and  $\hat{P}_{loss}(\hat{T}_j)$ ,

$$P_{loss}(T_j) = \alpha_P \cdot \hat{P}_{loss}(\hat{T}_j) \tag{6.6}$$

Let  $\hat{\mathbf{R}}_{eqC}$  be the matrix of the equivalent thermal resistances from the points of the 2D case temperature measurements to ambient estimated by (6.4) using  $\hat{P}_{loss}(\hat{T}_j)$  and  $\mathbf{T}_C$ . Then, according to (6.4) and (6.6), the relation between  $\hat{\mathbf{R}}_{eqC}$  and  $\mathbf{R}_{eqC}$  can be obtained:

$$\widehat{\mathbf{R}}_{eqC} = \frac{\mathbf{T}_c - T_a \cdot \mathbf{J}}{\widehat{P}_{loss}(\widehat{T}_j)} = \alpha_P \cdot \frac{\mathbf{T}_c - T_a \cdot \mathbf{J}}{P_{loss}(T_j)} = \alpha_P \cdot \mathbf{R}_{eqC}$$
(6.7)

According to (6.7),  $\hat{\mathbf{R}}_{eqC}$  is proportional to  $\mathbf{R}_{eqC}$  by a scalar  $\alpha_P$ . This indicates that other than using  $\mathbf{R}_{eqC}$ , it is equivalent to use  $\hat{\mathbf{R}}_{eqC}$  to obtain  $k_{hc}$  in (6.5) for monitoring the aging of the substrate solder without the need for using the  $P_{loss}(T_j)$ .

Moreover, since  $k_{hc}$  only changes with the aging process of the substrate solder but is independent of the operating condition of the IGBT module, the relationships between  $k_{hc}$  and  $\mathbf{R}_{eqC}$  and between  $k_{hc}$  and  $R_{thjc}$  can be easily characterized. For example, such relations can be obtained by recording the evolutions of  $k_{hc}$ ,  $\mathbf{R}_{eqC}$ , and  $R_{thjc}$  over time at a certain operating condition using an offline substrate solder aging test and stored in a lookup table. Then, according to  $k_{hc}$ , the values of  $\mathbf{R}_{eqC}$  and  $R_{thjc}$  can be obtained in real time for various operating conditions of the IGBT module even when a mixed mode aging process occurs in the IGBT module.

Once  $\mathbf{R}_{eqC}$  is known,  $\alpha_P$  can be calculated by using (6.7). Then, the power loss,  $P_{loss}$ , can be calculated from (6.6) using  $\hat{P}_{loss}$  and  $\alpha_P$ . In addition, according to the linear thermal behavior of the IGBT module,  $T_j$  and the corresponding change in the total power loss,  $\Delta P_{loss}(T_j)$ , during the aging process can be updated by using (6.8) and (6.9), respectively.

$$T_j = \alpha_P \cdot \hat{P}_{loss}(\hat{T}_j) \cdot (R_{thjc} + R_{thca}) + T_a$$
(6.8)

$$\Delta P_{loss}(T_j) = P_{loss}(T_j) - \hat{P}_{loss}(T_j)$$
(6.9)

The use of the updated  $T_j$  to recalculate  $\hat{P}_{loss}$  in (6.9) removes the power loss increase caused by the rise of the junction temperature from  $\Delta P_{loss}(T_j)$ , and the resulting  $\Delta P_{loss}(T_j)$ only reflects the aging of bond wires and emitter metallization. Then, the change of  $V_{CEon}$ at  $T_j$ ,  $\Delta V_{CEon}(T_j)$ , is obtained by using (6.10) to monitor the aging processes of bond wires and emitter metallization:

$$\Delta V_{CEon}(T_j) = \frac{\Delta P_{loss}(T_j)}{\hat{P}_{closs}(T_j)}$$
(6.10)

The changes of the key parameters for different aging mechanisms are summarized in Table 6.1, where  $\alpha_P$  may increase during the aging process of the substrate

solder because of the elevation of  $T_j$ , which increases the power loss generation. Table 6.1 clearly shows that by monitoring these parameters online, not only the aging of the IGBT module can be detected but also the aging mechanism(s) can be identified.

#### 6.2.4 Other Major Aging Mechanisms

Since the DBC layers have good heat spreading properties, the shrinking or alternation of the heat conduction area caused by a crack or void in the chip solder typically has a negligible impact on  $\mathbf{R}_{eqC}$  and, therefore, does not alter the nonuniformity of the case temperature distribution. However, if an aging mechanism causes an increase in  $P_{loss}$  but does not change  $\mathbf{R}_{eqC}$ , then according to the linear thermal behavior expressed by (6.7), the aging process can be monitored by the increase in ( $\mathbf{T}_C - T_a \cdot \mathbf{J}$ ) indirectly.

### 6.3 Implementation of the Proposed Method

A simple implementation of the proposed method is to place two thermal sensors, such as thermocouples, for each switch on the bottom surface of the baseplate of an IGBT module. One sensor should be placed at the hot spot, e.g., underneath an IGBT chip of the switch, to monitor the case temperature,  $T_{cchip}$ , which is most sensitive to the changes in  $P_{loss}(T_j)$ . The other sensor should be placed somewhere away from the hot spot to monitor the case temperature,  $T_{cside}$ . For example, the placement of thermal sensors for a three-phase inverter module having six IGBT switches is illustrated in Fig. 6.2. The use

Aging Mechanisms	Ploss	khc	Ø.P	
Bond wire, emitter	Increase	Constant	Increase	
Substrate solder	Rely on $T_j$	Increase	Rely on $T_j$	
Mixed mode aging	Increase	Increase	Increase	

Table 6.1: Changes of key parameters for different aging mechanisms.



Fig. 6.2: The placement of thermal sensors on the bottom surface of the baseplate of a three-phase inverter module having six IGBT switches.

of thermocouples to measure temperature was discussed in [65]. The temperature signals measured are compensated and analog-to-digital converted by using an integrated circuit (IC), such as MAX31855 [66].

For one IGBT switch, the ratio  $k_{cs}$  (a specific case of  $k_{hc}$ ) between the equivalent thermal resistances from the two selected points to ambient is used to evaluate the nonuniformity of the case temperature distribution, which is solely related to the alternation of the heat flow inside the IGBT module caused by the substrate solder aging:

$$k_{cs} = \frac{\hat{R}_{eqCchip}}{\hat{R}_{eqCside}} = \frac{R_{eqCchip}}{R_{eqCside}}$$
(6.11)

where  $\hat{R}_{eqCchip}$  and  $\hat{R}_{eqCside}$  are the equivalent thermal resistances from the two selected points to ambient, respectively, estimated from (6.7) using  $\hat{P}_{loss}(\hat{T}_j)$  and measured  $T_a$ ,  $T_{cchip}$ , and  $T_{cside}$ ;  $R_{eqCchip}$  and  $R_{eqCside}$  are the corresponding actual equivalent thermal resistances. Since the values of  $k_{cs}$  of all the six switches are independent of the operating condition of the IGBT module, it can be used for real-time monitoring of the substrate solder for various operating conditions of the IGBT module. When the IGBT module experiences a mixed mode aging process, the change of  $V_{CEon}$  over time can be obtained in real time according to the procedure described in Section 6.2.3 for monitoring the aging processes of bond wires, emitter metallization, etc. Since the case temperatures do not vary quickly, a short period steady state can be found in practice for the execution of the proposed method. The flowchart of the proposed method is shown in Fig. 6.3, where *d* is the duty ratio and *K* is typically in the range of 15-20.

The proposed method is capable of monitoring the major aging modes separately by separating the changes in case temperature caused by the increase of the power loss from those caused by the heat flow change during the aging process of the IGBT module. The method in [53] only uses a one point case temperature to detect the rise of the power loss for solder age monitoring. This, however, can be erroneous because an increase in the one point case temperature can be a result of an increase in the power loss or an alteration of the heat flow inside the IGBT module.

The scope of this work is monitoring the aging of IGBT modules, which does not consider the degradation of the thermal grease between the IGBT module and the cold plate. The issue of thermal grease degradation was discussed in [67], [68] and can be easily addressed by regular maintenance [67]. The inspection and maintenance for thermal grease is much easier than that for IGBT modules. Moreover, the recent development in thermal interface materials has dramatically improved the reliability of the thermal interface between the IGBT module and the cold plate [69]. With proper thermal grease maintenance, the potential impact of thermal grease degradation on the proposed method can be safely eliminated.



Fig. 6.3: The flowchart of the implementation of the aging monitoring method using two case temperatures.

## 6.4 Simulation Validation

The commercial IGBT module CM400DY-12NF made by Powerex (described in Chapter 3) was studied to validate the proposed case-temperature-based, real-time monitoring method for aging of IGBT modules. Two points at the bottom surface of the baseplate of the IGBT module, one underneath an IGBT chip and the other underneath the edge of the substrate solder layer, were selected for measuring the case temperature, as shown in Fig. 6.4.

The same FEA model of the test IGBT module built in Autodesk Simulation Multiphysics, described in Chapter 3, was used. The temperature of the pipes was set to be the inlet coolant temperature of 20 °C, i.e.,  $T_a = 20$  °C. The IGBT module was operated as a leg in a two-level three-phase inverter with the following operating conditions: the DC-link voltage was 400 V,  $I_c = 400$  A,  $f_{sw} = 5$  kHz, the power factor PF



Fig. 6.4: The test IGBT module with the positions of the two thermal sensors placed on the bottom surface of the baseplate (top view from the upper side of the module).

= 0.8, a unity modulation index, and 60 Hz line frequency.

When the IGBT module was healthy, the initial junction temperature was 75.2 °C; and the corresponding initial power losses of the IGBT chips and the FWD chips in a switch were calculated to be 227.63 W and 42.52 W, respectively, using the equations in [62]. The initial switching loss of the IGBT switch was 69.48 W, which increased during the aging process as the junction temperature increased. Then, based on (6.6), (6.9), and (6.10), the increase in  $\alpha_P$  related to the increase in  $V_{CEon}$  could be obtained. The losses of the FWD elevated  $T_{cchip}$  and  $T_{cside}$  by approximately 4.7 °C and 3.4 °C, respectively, which were almost constant throughout the aging process of the IGBT module due to a relatively long distance between the FWD and the two selected points on the baseplate and the relatively small values of the FWD losses. In practice, the case temperature increases caused by the FWD losses should be removed to perform the proposed method, which can be easily implemented by applying superposition.

The aging processes of the substrate solder, bond wires, and emitter metallization of the IGBT module were simulated. Snapshots were taken in the following four scenarios to show the aging development in the IGBT module: 1) healthy condition, 2) a minor aging in the substrate solder with a 2 mm crack, 3) an intermediate aging of the substrate solder with a 4 mm crack and an aging of bond wires and emitter metallization mimicked by a 10% increase in  $V_{CEon}$ , and 4) a dangerous aging status with a 6 mm crack in the substrate solder and a 20% increase in  $V_{CEon}$ . The substrate solder cracks were assumed to be uniform from the edge to the center. The measured  $T_{cchip}$  and  $T_{cside}$  and the calculated  $T_j$ ,  $R_{thjc}$ , and  $P_{loss}$  in the four scenarios during the simulated aging process of the IGBT module are shown in Fig. 6.5.



Fig. 6.5: Evolution of parameters during the mixed mode aging process of the IGBT module: (a)  $T_{cchip}$ ,  $T_{cside}$ , and  $T_j$ ; (b)  $P_{loss}(T_j)$ ; and (c)  $R_{thjc}$ .

The value of  $R_{thjc}$  increased during the development of a substrate solder crack. The power loss increased noticeably when  $V_{CEon}$  increased 10% and 20%. The value of  $T_j$  first increased slightly due to an increase in  $T_{cchip}$  caused by the substrate solder crack in Scenario 2, then increased dramatically in Scenarios 3 and 4 due to the increases in  $P_{loss}$ ,  $R_{thjc}$ , and  $T_{cchip}$ . The value of  $T_{cchip}$  increased due to the increasing nonuniformity of the case temperature distribution and the increase in  $P_{loss}$ . Note that  $T_{cside}$  decreased first due to the substrate solder crack and then increased due to the increase in  $P_{loss}$ .

The proposed method only needs additional measurements of  $T_{cchip}$  and  $T_{cside}$  to evaluate the health condition of the IGBT module. The values of  $\hat{R}_{eqCchip}$  and  $\hat{R}_{eqCside}$  were estimated by using (6.7) with the information of  $\hat{P}_{loss}$ ,  $T_{cchip}$ , and  $T_{cside}$ ; and the results are shown in Fig. 6.6(a). Then,  $k_{cs}$  was calculated by using (6.11); and the result is shown in Fig. 6.6(b). The dip and increase in  $\hat{R}_{eqCside}$  indicated a mixed aging model. The value of  $k_{cs}$ , however, increased monotonically along with the development of the substrate solder crack and increased 16% from the healthy case when the crack became 6 mm.

Table 6.2 characterizes the relationships between  $k_{cs}$ ,  $R_{eqCchip}$ ,  $R_{eqCside}$ , and  $R_{thjc}$ during the aging process. The table was built based on the simulation data from a single substrate solder aging (crack) process when a 200-A DC current was applied to a switch of the IGBT module (i.e.,  $I_c = 200$  A). From Table 6.2, the value of  $R_{thjc}$  can be found based on the value of  $k_{cs}$ , which matches the evolution of  $R_{thjc}$  in Fig. 6.5.



Fig. 6.6: Evolution of the estimated parameters during the mixed mode aging process of the IGBT module: (a)  $\hat{R}_{eqCchip}$  and  $\hat{R}_{eqCside}$ ; (b)  $k_{cs}$ .

Table 6.2:	Lookup	table for	$k_{cs}$ and	thermal	resistances	for	solder	aging	monitoring.
------------	--------	-----------	--------------	---------	-------------	-----	--------	-------	-------------

$k_{cs}$	1.53	1.59	1.71	1.78
$R_{eqCchip}$ (°C/W)	0.148	0.150	0.154	0.157
$R_{eqCside}$ (°C/W)	0.097	0.094	0.090	0.088
$R_{thjc}$ (°C/W)	0.0749	0.0749	0.0759	0.0920



Fig. 6.7: Calculated  $\alpha_P$  during the aging process of the IGBT module.

By using  $R_{eqCchip}$  and  $R_{eqCside}$  and applying (6.7), the value of  $\alpha_P$  was calculated, as shown in Fig. 6.7. The junction temperature  $T_j$  could be updated during the aging process by (6.8) using  $R_{thjc}$  and  $\alpha_P \cdot \hat{P}_{loss}(\hat{T}_j)$ , where  $\hat{T}_j = 75.2$  °C was the initial junction temperature when the IGBT module was healthy. In Scenarios 3 and 4, the updated values of  $T_j$  were 81.4 °C and 91.5 °C, respectively; the values of  $\hat{P}_{loss}(T_j)$  were 230.46 W (the conduction loss was 159.73 W) and 236.49 W (the conduction loss was 162.89 W), respectively. By using (6.9) and (6.10), the percentage increases in  $V_{CEon}$  due to bond wire and emitter metallization aging were correctly detected. Since  $k_{cs}$ ,  $R_{eqCchip}$ ,  $R_{eqCside}$ , and  $R_{thjc}$  are independent of the operating condition because of the linear thermal behavior of the IGBT module, The lookup table (e.g., Table 6.2) obtained at a specific operating condition can be used for condition monitoring of the IGBT module for various operating conditions. This feature is validated by experimental results in the next section.

## 6.5 Experimental Validation

Experimental studies were conducted to further validate the proposed method. Fig. 6.8 illustrates the experimental setup, which consisted of an oscilloscope for steady and



(b)

Fig. 6.8: Experimental setup for validation of the proposed method: (a) the schematic and (b) the actual setup.

transient electrical signal acquisition, such as  $I_c$  and  $V_{CE}$ ; a DC source for generating gate signals for the test IGBT module; a programmable high-power DC source for providing test currents; a liquid recirculating chiller for providing the inlet coolant at 20 °C for cooling the IGBT module; a National Instruments (NI) data acquisition system (with a measurement resolution of 0.01 °C) for acquiring  $T_j$ ,  $T_{cchip}$ , and  $T_{cside}$ , which were measured by using precise fine wire (0.13 mm diameter) thermocouples, Omega 5TC-TT-K-36-36, with insulation; and a control system developed in LabVIEW<sup>TM</sup> operating on a laboratory computer. The thermocouples were recommended by the manufacturer of the IGBT module to measure the case temperature for calculating the module's thermal resistance [70]. Shallow grooves were carved on the bottom surface of the baseplate and filled with thermal grease for the installation of the thermocouples.

#### 6.5.1 Validation of the Linear Thermal Behavior of the IGBT Module

To validate the linear thermal behavior of the IGBT module, which is the foundation of the proposed method, a set of tests were performed by applying different DC currents to the IGBT module. In each test, a DC current with an amplitude between 80 A and 150 A was applied to the IGBT module to heat it until a steady state was reached. Different current amplitudes were used to emulate variations in the power loss

$I_{c}(\mathbf{A})$	$V_{CE}$ (V)	$T_{Cchip}$ (°C)	T <sub>Cside</sub> (°C)	$T_{cool}$ (°C)
80	1.16	53.3	50.5	20
100	1.197	63.1	59.5	20
120	1.227	73.1	68.6	20
150	1.308	90	84.2	20

Table 6.3: The measured temperatures at different operation conditions.
and junction temperature of the IGBT. The  $R_{thja}$  was deliberately increased by mounting the module on the cold plate with a low torque to realize higher  $T_j$  variations. This facilitated examination of the linearity of the thermal behavior of the IGBT module in a wide operating temperature range. The values of  $T_{cchip}$ ,  $T_{cside}$ , and coolant (ambient) temperature  $T_a$  at different currents were recorded in Table 6.3. Then,  $R_{eqCchip}$ ,  $R_{eqCside}$ , and  $k_{cs}$  were calculated using these data. Although the power loss and junction temperature varied significantly at different currents, the values of  $R_{eqCchip}$ ,  $R_{eqCside}$ , and kwere almost constant, as shown in Fig. 6.9. The maximum variations of  $R_{eqCchip}$ ,  $R_{eqCside}$ , respectively. Such small variations can be neglected. These results validated the linearity of the thermal behavior of the IGBT module over a wide range of junction temperatures. The value of  $k_{cs}$  was almost constant at different power loss conditions if the health condition of the solder was unchanged.



Fig. 6.9: The values of  $R_{eqCchip}$ ,  $R_{eqCside}$  and  $k_{cs}$  at different operating conditions.

#### 6.5.2 Solder Aging Monitoring

After the linear thermal behavior of the IGBT module in a specific health condition was validated, the values of  $k_{cs}$  and  $\alpha_P$  were estimated using the proposed method to monitor the solder aging and mixed mode aging of solder and bond wires.

First, a thermal cycling test was designed to accelerate the aging process of the substrate solder, where the thermal cycles were excited by the power loss in the IGBT module. To maximize the temperature variation in the substrate solder layer while keeping  $T_j$  below 125 °C, the allowed maximum  $T_j$  specified by the manufacturer, the IGBT module was air cooled and powered by a relatively small current of 80 A. In this case,  $R_{thca}$  was much larger than  $R_{thjc}$ , making the case temperature close to the junction temperature. In each thermal cycle, the 80 A current was applied to the IGBT module for 3 minutes to gradually heat it up. Then, the module was cooled down in the next 15 minutes. The maximum  $T_j$  per thermal cycle was 105 °C. The variation of  $T_{cside}$  in one thermal cycle was about 70 °C, which was only 10 °C less than that of  $T_j$ .

The thermal cycling test was interrupted after certain cycles to inspect the changes in  $T_{cchip}$ ,  $T_{cside}$ ,  $T_j$ , and  $R_{thjc}$ . The value of  $k_{cs}$  was calculated using the measured case temperatures. During the inspection test, the IGBT module was remounted on the cold plate with the mounting torque recommended by the datasheet and was powered by a 200 A DC current. No change in  $V_{CEon}$  was observed. The test results are shown in Fig. 6.10 and are consistent with the simulation results. After 3,300 thermal cycles,  $R_{thjc}$  only increased by 4.5%; and  $T_{cchip}$ ,  $T_{cside}$ , and  $T_j$  decreased slightly due to a 0.7 °C decrease in the coolant temperature. After that, the difference between  $T_{cchip}$  and  $T_{cside}$ , ( $T_{cchip}$ - $T_{cside}$ ), increased from 13 °C at about 3,300 thermal cycles to 15 °C when  $R_{thjc}$  increased 12.2%



Fig. 6.10: Evolution of parameters during thermal cycling of the IGBT module: (a) measured  $T_{cchip}$ ,  $T_{cside}$ , and  $T_j$ ; (b)  $R_{thjc}$ .

at about 5,100 thermal cycles, and finally increased to 19.8 °C when  $R_{thjc}$  increased 18.6% at the end of the test. The nonuniformity of the case temperature distribution grew significantly during the aging process of the substrate solder.

The value of  $k_{cs}$  was estimated during the thermal cycling test, as shown in Fig. 6.11. The trend of change of  $k_{cs}$  obtained from the experiment was similar to that in the simulation. Compared to the first thermal cycle, the value of  $k_{cs}$  only increased 0.2% at



Fig. 6.11: Evolution of *k* during the thermal cycling test of the IGBT module.

the 3,300 thermal cycle, indicating that the IGBT module was healthy. However, the value of  $k_{cs}$  increased 6.3% at the 5,100 thermal cycle and 25% at the end of the test, which indicates that the IGBT module was in a medium aging and a severe aging condition, respectively. The latter also indicates that maintenance is needed for the power converter. The experimental results proved that the substrate solder aging of the IGBT module can be effectively monitored by the proposed method using  $T_{cchip}$  and  $T_{cside}$ .

#### 6.5.3 Mixed Mode Monitoring of Aging

After the thermal cycling test, a power cycling test was performed to accelerate the aging process of the bond wires of the same IGBT module mounted onto the cold plate. In each power cycle, a 400 A current was applied to the IGBT module for 5 seconds, followed by 15 seconds to cool down the module. The variation of  $T_j$  in a power cycle was about 100 °C. The value of  $V_{CEon}$  at the same junction temperature of 90 °C was acquired in each power cycle. The evolution of  $V_{CEon}$  at 400 A during the power cycling test is shown in Fig. 6.12. The value of  $V_{CEon}$  increased 10% at the end of the power cycling test. No increase in  $R_{thjc}$  was observed. The power cycling test was interrupted after certain cycles to inspect the changes in  $\alpha_P$  and  $k_{cs}$  by applying 100 A and 200 A DC currents to the IGBT module. The values of  $\alpha_P$  and  $k_{cs}$  were calculated during the steady state after the DC currents were applied for the mixed mode aging. As shown in Fig. 6.13, The value of  $\alpha_P$  increased from 1 at the beginning of the power cycling test to 1.12 at 200 A and to 1.09 at 100 A at the end of the power cycling test. The value of  $\alpha_P$ calculated at 200 A was larger than that at 100 A due to the temperature effect on the power loss generation. The values of  $k_{cs}$  were around 1.82 at different currents and different bond wire health conditions. The small variation in  $k_{cs}$  was caused by coolant temperature variations and thermocouple measurement errors.

The experimental results proved that: 1) it is effective to use  $k_{cs}$  for monitoring the aging of substrate solder, and  $k_{cs}$  is independent of the health condition of the bond



Fig. 6.12: Evolution of  $V_{CEon}$  during the power cycling test of the IGBT module.

wires; and 2)  $\alpha_P$  is sensitive to changes in  $V_{CEon}$  and, therefore, is an effective precursor for monitoring the aging of bond wires and emitter metallization.



Fig. 6.13: Evolution of parameters during the power cycling test of the IGBT module: (a)  $\alpha_P$  and (b)  $k_{cs}$ .

### 6.6 Summary

This dissertation has proposed a real-time method for monitoring the aging of IGBT modules using 2D case temperatures. The proposed method is effective for monitoring major aging processes in an IGBT module, such as bond wire liftoff, metallization aging, and substrate solder aging. A simple implementation of the proposed method using two temperature sensors for each switch was validated by simulation and experimental studies on a commercial IGBT module. Results have shown that a mix of the major aging processes can be easily detected and separated by monitoring the case temperatures at two selected points. As the case temperature is relatively stable and easy to monitor, the proposed method can be easily implemented for monitoring the aging of IGBT modules in power converters. It is recommended that the calculation of  $k_{cs}$  and  $\alpha_P$  using the proposed method be performed when the case temperature reaches a steady state. However, when the power converter experiences a transient load, the proposed method can also be performed by averaging the case temperature measurements over a relatively long period to filter out the variations caused by the load transient.

Table 6.4 compares the proposed method with three other methods in the literature. It reveals that the proposed method achieved the best performance for all of the three metrics. For example, the proposed method is capable of monitoring all of the three major aging mechanisms and detecting which aging process or processes occur. These cannot be achieved by any of the three other methods. Moreover, compared to the first two methods in the table, the implementation of the proposed method has lower hardware complexity and does not affect power converter operation.

Methods	Aging Mechanisms That Can Be Detected	Hardware Complexity	Affect Converter Operation
Using V <sub>CEon</sub>	Bond wire, emitter metallization	High	Possible
Using TSEP	Any mechanisms causing $T_j$ elevation	High	Yes
Using $(T_c - T_a)$ [53]	Solder	Low	No
Proposed method	Bond wire, emitter metallization, substrate solder	Low	No

Table 6.4: Comparison of aging monitoring methods for IGBT modules.

# Chapter 7. Frequency-Domain Transient Temperature Estimation and Aging Analysis for Weak Points

This chapter presents a new frequency-domain thermal model for online junction temperature estimation of IGBT modules [56]. This model characterizes the thermal behavior of an IGBT module by a linear time invariant (LTI) system. The frequency response at junction is obtained by applying the fast Fourier transform (FFT) to the time derivative of the transient thermal impedance from junction to a reference position of the IGBT module. The junction temperature of the IGBT is then estimated using the frequency responses at junction and the frequency spectra of heat sources of the IGBT module. This model can be extended to estimate temperature responses at different weak points and analyze the aging modes of the IGBT module under complex working conditions [25]. By examining the frequency spectra of the power losses of an IGBT at an operating condition and the frequency responses of the temperatures at weak points, the information on bottleneck weak point(s) can be obtain. Simulation results show: 1) the proposed method is accurate and computationally efficient for estimating junction temperature and temperatures at the weak points of IGBT modules in both steady-state and transient loading conditions; and 2) the proposed method can reveal that the power losses at different loading conditions with different frequency spectra stress the weak points differently.

#### 7.1 Heat Equation for IGBT Modules

The IGBTs with the "standard" power module packaging [17] are considered herein. During the operation of an IGBT module, the heat is generated at the chips and spreads through multiple layers to the bottom of the baseplate, which is cooled by a heat sink or cold plate. The heat conduction in the heat diffusion layers is governed by Fourier's law. By assuming that the materials are isotropic and neglecting the temperature effects of the materials' thermal properties, a differential form of heat equation governing the heat diffusion in a certain layer in the rectangular coordinate system is given by

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} + \frac{\partial^2 T}{\partial z^2} = \frac{1}{\alpha} \frac{\partial T}{\partial t}$$
(7.1)

where *T* is the temperature and  $\alpha$  is the thermal diffusivity of the material of the layer. The thermal dynamics of the overall IGBT module can be modeled by equations similar to (7.1) for all of the layers constrained by the boundary conditions between different layers.

The exact solution to (7.1) can be very complex if the module geometry and boundary conditions are taken into account.

# 7.2 Frequency Domain Junction Temperature Estimation

Since (1) is linear, the dynamical thermal behavior of an IGBT module can be characterized by an LTI system by neglecting the cooling system performance variations. In this section, the relationship between an IGBT's power loss,  $P_{loss}(t)$ , and the temperature rise from junction to case,  $\Delta T_{jc}(t)$ , is characterized by a high-order LTI system,  $h_{pj}(t)$ , which represents the time-domain impulse response of  $\Delta T_{jc}(t)$  excited by the heat source  $P_{loss}(t)$ . The junction temperature,  $T_j(t)$ , is then given by

$$T_j(t) = \Delta T_{jc}(t) + T_c(t) = P_{loss}(t) * h_{pj}(t) + T_c(t)$$
(7.2)

In this section,  $T_j(t)$  is estimated in the frequency domain using the Fourier transform technique to avoid determination of the explicit expression for  $h_{pj}(t)$  and calculation for the convolution between  $p_{loss}(t)$  and  $h_{pj}(t)$ , as follows:

$$T_{j}(t) = P_{loss}(t) * h_{pj}(t) + T_{c}(t) = \mathcal{F}^{-1} \left( P_{loss}(e^{jw}) \cdot H_{pj}(e^{jw}) \right) + T_{c}(t)$$
(7.3)

where  $P_{loss}(e^{j\omega})$  is the frequency spectrum of  $P_{loss}(t)$  obtained by using the Fourier transform,  $H_{pj}(e^{j\omega})$  is the frequency response of  $h_{pj}(t)$  obtained by using the Fourier transform, and  $\mathcal{F}^{-1}$  is the inverse Fourier transform. The  $h_{pj}(t)$  can be determined by taking the time derivative of the transient thermal impedance  $Z_{thjc}(t)$  from the junction to the case, i.e.,

$$h_{pj}(t) = dZ_{thjc}(t)/dt \tag{7.4}$$

The values of  $Z_{thjc}(t)$  can easily be obtained from FEA simulations or experiments.

For a compact IGBT module consisting of multiple heat sources, the superposition principle can be applied to characterize the thermal behavior of the whole IGBT module to obtain the junction temperatures at multiple chips of interest. The matrix form of the proposed method for obtaining the junction temperatures of N chips of interest for an IGBT module with M heat sources is given by

$$\begin{bmatrix} T_{j_1} \\ T_{j_2} \\ \vdots \\ T_{j_N} \end{bmatrix} = \mathcal{F}^{-1} \left( \begin{bmatrix} H_{11} & H_{12} & \dots & H_{1M} \\ H_{21} & H_{22} & \dots & H_{2M} \\ \vdots & \ddots & \vdots \\ H_{N1} & H_{N1} & \dots & H_{NM} \end{bmatrix} \begin{bmatrix} P_1 \\ P_2 \\ \vdots \\ P_M \end{bmatrix} \right) + T_c \cdot \mathbf{1}_{1 \times N}$$
(7.5)

where  $H_{nm}$  ( $n = 1, \dots, N$  and  $m = 1, \dots, M$ ) is the frequency response of the temperature rise from the *n*th junction to the case excited by the *m*th source  $P_m$ , and  $\mathbf{1}_{1\times N}$  is a *N*dimensional column vector with all elements being one. The value of  $H_{nm}$  can be determined by the Fourier transform of the time derivative of  $Z_{nm}(t)$ , which is the equivalent transient thermal impedance from the *n*th junction to case when only the *m*th source is powered. The value of  $Z_{nm}(t)$  can be obtained from FEA simulations or experiments.

In digital system applications, the discrete time form of (7.3) is given by

$$T_j[k] = P_{loss}[k] * h_{pj}[k] \cdot T_s + T_c[k] = \mathcal{F}^{-1} \left( P_{loss}(\Omega) \cdot H_{pj}(\Omega) \right) \cdot T_s + T_c[k]$$
(7.6)

where k is the index of the discrete time sequence,  $T_s$  is the sampling period,  $P_{loss}(\Omega)$  is the discrete Fourier transform (DFT) of  $P_{loss}[k]$ ,  $H_{pj}(\Omega)$  is the frequency response of  $h_{pj}[k]$ , and  $h_{pj}[k]$  can be calculated by

$$h_{pj}[k] = (Z_{jc}[k+1] - Z_{jc}[k])/T_s$$
(7.7)

The DFT can be efficiently computed using the FFT. The flowchart of the proposed method is illustrated in Fig. 7.1.

The proposed method has the following advantages.

1) Highly Accurate: The accuracy of the proposed method relies on the accuracy of  $h_{pj}(t)$  of the IGBT module, which can be accurately estimated offline. Therefore, the proposed method is capable of approximating the accurate numerical thermal model obtained from FEA simulations or the experiment measurements online with good precision.

2) Computationally Efficient: The computational cost of the proposed method depends on the sampling frequency of the quantities in (7.5). According to the Nyquist-Shannon sampling theorem,  $f_{sw}$  of the power losses in (7.5) should be more than twice the highest harmonics of the power losses. According to the power loss calculation equations in [77], the highest harmonics of the power losses in an inverter are just several times higher than the line frequency, if the power losses are averaged over each switching period. Thus, the proposed method does not require a high sampling frequency. The proposed method has a computational complexity of  $O(K_{sp} \times \log_2 N_l)$ , where  $K_{sp}$  is the number of the time-domain data samples of the power losses; and  $N_l$  is the order of the LTI system. The value of  $N_l$  is the product of the sampling frequency in Hz, e.g., 1,000



Fig. 7.1: The flowchart of the proposed method.

Hz, and the settling time in second to the steady state (typically 5-10 s) of the transient response of the time domain thermal impedance. The computational complexity of the proposed method is on the same order as that of a TEC model  $O(K_{sp} \times N_{rc})$ , where  $N_{rc}$  is the order of the TEC. Therefore, the proposed method is computationally efficient for online junction temperature estimation.

*3) Easy to Build:* The derivation of a high-fidelity, analytical thermal model from physical principles or the extraction of a high-fidelity, low-order numerical thermal model from a high-order numerical thermal model of an IGBT module usually involves significant mathematical complexity and difficulty, especially when the target IGBT module has high geometric complexity and multichip thermal cross-coupling. The proposed method does not have any problem with mathematical complexity or difficulty.

# 7.3 Frequency Domain Aging Model Analysis

#### 7.3.1 Thermomechanical Stresses

As stated in Chapter 1, the major aging mechanisms observed in power devices are mainly due to thermomechanical stresses experienced by the packaging materials, which are caused by the mismatch of the CTEs between different materials. The CTEs of copper and aluminum are much larger than those of silicon and ceramic. Therefore, during temperature swings, thermomechanical stresses are generated between the interfaces of different layers made of different materials, which results in the aging of weak points. The aging of bond wires is a result of either the shear stress generated between the bond pad and the wire or by the repeated flexure of the wires [10] during temperature swings. The aging of a solder layer in an IGBT module is mainly due to the large mismatch of the CTEs of the two adjacent layers and temperature swings combined with the dimensions of the layers. The strain,  $\varepsilon_{mismatch}$ , at the interface between two different materials caused by the thermal mismatch of their CTEs is calculated by

$$\varepsilon_{mismatch} \approx \Delta \alpha_{CTE} \cdot \Delta T_{\varepsilon}$$
 (7.7)

where  $\Delta \alpha_{CTE}$  is the difference between the CTEs of the two materials, and  $\Delta T_{\varepsilon}$  is the temperature change from the temperature where the strain is zero [8]. By substituting (7.7) into the Coffin-Manson equation for the low-cycle fatigue (dominated by the plastic strain) and the Basquin's equation for the high-cycle fatigue (dominated by the elastic strain) [9], the lifetime of the interface between the two materials under cyclic thermomechanical stresses has an inverse exponential relationship with  $\Delta T$  as follows:

$$\frac{\Delta \varepsilon_p}{2} = \varepsilon_f' (2N_f)^{c_f} \to N_f \propto (\Delta \varepsilon_p)^{-c_f} \propto (\Delta T)^{-c_f}$$
(7.8)

$$\sigma_a = E \frac{\Delta \varepsilon_e}{2} = \sigma_f' (2N_f)^{b_f} \to N_f \propto (\Delta \varepsilon_p)^{-b_f} \propto (\Delta T)^{-b_f}$$
(7.9)

where  $\varepsilon_p$  is the plastic strain,  $\varepsilon'_f$  is the fatigue ductility coefficient,  $c_f$  is the fatigue ductility exponent,  $\varepsilon_e$  is the elastic strain,  $\sigma_a$  is the alternating stress, *E* is Young's modulus,  $\sigma'_f$  is the fatigue strength coefficient, and  $b_f$  is the fatigue strength exponent. It should be noted that  $\Delta T$  is the local temperature swings at the weak points. According to (7.8) and (7.9), the bond wire and large area solder lifetimes are related to the local temperature swings.

#### 7.3.2 Aging Mode Analysis

The existing device aging analysis usually only uses the junction temperature information and a lifetime mode, such as (1.4). However, the junction temperature is much more sensitive to a transient power loss than the temperatures at other weak points.

IGBT modules with similar junction temperature records might have quite different lifetimes. For example, a 10 °C swing in the junction temperature can be caused by a gradual or sudden change in power loss. However, for the temperature at the free edge of the substrate solder, a negligible swing is observed when a very short pulse power loss is applied to the IGBT. These facts indicate that an IGBT module might experience quite different stresses in different loading conditions when the records of  $\Delta T_j$  are very close. In other words, to achieve an accurate lifetime prediction, the power loss profiles at the loading conditions and the corresponding bottleneck weak point(s) should be analyzed appropriately. Thus, a study on the temperature swings at the weak points when an arbitrary loading condition is applied to the power device is highly desirable.

Several points in an IGBT module are used to study temperature swings at weak points, including the middle of junction ( $P_{jm}$ ), the free edge of the chip solder layer ( $P_{ce}$ ), and the free edge of the substrate solder ( $P_{se}$ ), as illustrated in Fig. 7.2. The points are chosen due to the fact that bonds usually fail from the center to the edge of the chip [79] and a crack usually starts at the free edge of a large-area solder layer in electronic



Fig. 7.2: Selected points and heat flows inside an IGBT module.

packages [72]. During the operation of an IGBT, the heat can be assumed to be generated at the top surface of a chip and to spread down through different layers to the bottom of the baseplate, which is cooled by a heat sink or a cold plate. The temperature changes at the selected points are complex due to the high complexity of the module geometry and the effects of heat storage and spreading in layers. For example,  $P_{jm}$  is the location where most heat is generated. Therefore, the temperature at  $P_{jm}$  is sensitive to the changes in power loss, while the temperature at  $P_{se}$  performs a "low-pass filtered" thermal response due to its distance to the heat source and the heat storage effect in the DBC.

The method used to thoroughly study temperature swings at weak points of an IGBT module in general operating conditions is fairly straightforward—the frequency spectra of the local thermal responses are analyzed. In this section, the thermal analysis method proposed in Section 7.2 is extended to estimate the transient temperature at the weak points. The proposed method is highly accurate and can be used for any loading conditions.

The proposed method can reveal that the power losses at the loading conditions with different frequency spectra stress the weak points differently. As stated early in this section, due to the difference in geometry, the temperature at  $P_{se}$  has a different frequency response from that at  $P_{jm}$ . The IGBT module can undergo a power cycling which stresses bond wires and chip solder or a power cycling mixed with a thermal cycling which stresses all of the weak points. An example of the temperatures at  $P_{jm}$  and  $P_{se}$  under two different loading conditions is illustrated in Fig. 7.3. Since the power loss is nonnegative, the DC components of the power losses are large. Loading 1, which has rich high-frequency components, may not severely stress the substrate solder. However, the

reliability of bond wires might deserve more consideration. For Loading 2, the lifetime prediction for the IGBT module should consider the aging of both weak points.

It is difficult to define an arbitrary working cycle in the time domain [80]. However, by examining the frequency spectrum of the power loss at a loading condition, the information on the aging mode(s) of the IGBT can be obtained. Moreover, the analysis in the frequency domain can help classify the loads in terms of the aging modes of an IGBT. Note that a quantitative lifetime cannot be directly calculated from the power loss spectrum. This is because the Fourier transform of the power loss at a loading condition is a linear decomposition of temperature excitations, and the lifetime models are exponentially dependent on the temperatures at the weak points. However, the proposed method can help determine the bottleneck weak point(s) of an IGBT module under complex operating conditions by analyzing the frequency spectra of the power



Fig. 7.3: Temperature information of  $P_{jm}$  and  $P_{se}$  under two different loading conditions.

losses at the loading conditions.

# 7.4 Simulation Study

In this chapter, the simulation study consists of three parts: 1) validate the accuracy of the frequency domain transient junction temperature estimation method; 2) examine the differences between temperature responses at different weak points; and 3) preliminarily verify the frequency domain aging mode analysis method by comparing time domain lifetime predictions of the weak points.

#### 7.4.1 Frequency Domain Junction Temperature Estimation

Simulation studies were performed to validate the proposed frequency domain junction temperature estimation method against an FEA thermal model built in Autodesk Simulation Multiphysics, i.e., the reference model, and the 4<sup>th</sup> order TEC model provided by the manufacturer [81] for junction temperature estimation of the commercial IGBT module, CM400DY-12NF. Note that the FEA used in this study is different from the FEA thermal model in Chapter 3. In the FEA thermal model used in this study, a fine mesh was used for the chips, solder interfaces, and DBC stack; while a relatively coarse mesh with 7551 3D elements was used for the baseplate. The total mesh of the FEA thermal model of the IGBT module has 25,955 3D elements and 39,090 nodes. It was assumed that the power losses were generated inside the chips close to their top surfaces [35]. Therefore, the calculated power losses could be converted to heat fluxes applied at the top surfaces of the chips [23]. The cooling for the IGBT module was simplified to heat convection at the bottom surface of the baseplate with a constant heat convection rate of 10,000 W/(m<sup>2</sup>·°C). The ambient temperature was 65 °C. The IGBT module was

operated as a leg in a two-level three-phase inverter. The case temperature was measured at the point specified on the outline drawing and circuit diagram on the datasheet [8] for the IGBT module.

The transient thermal impedances from the junction of the IGBT chip to the case,  $Z_{thjcQQ}(t)$  and  $Z_{thjcQD}(t)$ , which were obtained when the IGBT chip was powered alone and its parallel FWD was powered alone, respectively, are shown in Fig. 7.4. The settling times to the steady state of  $Z_{thjcQQ}(t)$  and  $Z_{thjcQD}(t)$  were 5 s. The  $h_{QQ}(t)$  and  $h_{QD}(t)$  in Fig. 7.4, which were determined by taking the time derivatives of  $Z_{thjcQQ}(t)$  and  $Z_{thjcQD}(t)$ , respectively, were the impulse responses of the LTI systems characterizing the relationships between the IGBT junction temperature and the power losses generated by the IGBT and the FWD, respectively. The sampling frequency was 1 kHz for the proposed method. Therefore, the order of the LTI systems used in the proposed method was 5,000.

The same transient load in Chapter 4, which mimicked the starting process of an electric motor in an HEV, was applied to the inverter to evaluate the accuracy of the proposed method for online junction temperature estimation. In a TEC model, since the TECs from the junction to the case of the IGBTs and the FWDs were only interconnected at the heat sink, the thermal cross-coupling effects between chips in an IGBT module can only be partially addressed [82]. Thus, for fair comparison between different methods, the thermal couplings between IGBTs and FWDs were neglected in all of the three models. The power losses generated by the chips were calculated based on the simulated currents as well as the conduction voltage-current curve and switching loss-current curve provided by the datasheet of the IGBT module [8].



Fig. 7.4: The transient thermal impedance responses from the junction of the IGBT chip to the case and the corresponding LTI system responses: (a)  $Z_{thjcQQ}(t)$ ; (b)  $Z_{thjcQD}(t)$ ; (c)  $H_{QQ}(t)$ ; and (d)  $H_{QD}(t)$ .

Fig. 7.5(a) compares the values of the junction temperature,  $T_j(t)$ , estimated by the proposed model, the TEC, and the FEA method. The value of  $T_j(t)$  fluctuates in the range of 10 ~ 15 °C. The value of  $T_j(t)$  obtained from the proposed method matches that of the FEA method better than the TEC. As shown in Fig. 7.5(b), the errors of the junction temperature estimated by the proposed method with respect to the FEA method at its



Fig. 7.5: Comparison of  $T_j(t)$  estimated by the three methods when the inverter was operated with a transient load: (a) junction temperature and (b) error at peaks and valleys with respect to the FEA method.

peaks and valleys are almost less than 0.2 °C with the maximum of 0.6 °C at 0.28 s during the transient load. On the other hand, the errors of the TEC model are between 1 °C and 1.6 °C at the peaks and between 0.6 °C and 1.1 °C at the valleys. The results showed that in general, the junction temperature estimation error of the proposed method was less than 20% of that of the TEC model. Moreover, when the line frequency of the inverter increased, the relative error of  $T_j(t)$  obtained from the TEC model with respect to its amplitude of variations in one cycle increased because of more dynamic changes in the power loss. However, the increase of the inverter line frequency did not affect the proposed method, as long as the highest harmonic of the power loss was lower than half of the sampling frequency of the power loss.

The proposed method was further examined by comparing it with the FEA method at a constant operating point of the converter: 400 V DC-link voltage, 400 A output current, power factor of 0.8, unity modulation index, 5 kHz switching frequency, and 60 Hz line frequency. In this study, the thermal couplings between the chips of IGBTs and FWDs were considered in both models to further examine the capability of the proposed method for handling the thermal coupling effects. The results in Fig. 7.6 show a negligible difference between the proposed method and the FEA method because the relative error of the junction temperature estimated by the proposed method.



Fig. 7.6: Comparison of *T<sub>j</sub>(t)* estimated by the proposed method and the FEA method when the inverter was operated at a constant operating condition:
(a) junction temperature and (b) relative error at peaks and valleys with respect to the FEA method.

# 7.4.2 Illustration of Temperature Reponses at Different Weak Points

For simplicity without loss of generality, temperature swings of a "general" single-chip symmetric IGBT module (Fig. 7.7) were studied by using the proposed frequency-domain analysis method. For multichip IGBT modules, the temperatures at the points of interest were estimated by applying superposition of the thermal responses at the points of interest to the excitations of multiple heat sources. The size of the chip was

 $10 \times 10 \times 0.25$  mm. The geometry and material composition of the substrate were the same as the "standard" high power IGBT module with AlN ceramic and copper baseplate. The size of the baseplate was  $18 \times 18 \times 4$  mm. The cooling system was simplified to a heat convection surface at the bottom of the baseplate with the ambient temperate being set to 0 °C.

As Fig. 7.8 shows, the transient thermal impedance from  $P_{se}$  to ambient, which was derived from an FEA simulation using a step-change input power, was distinctive from those from  $P_{jm}$  and  $P_{ce}$  to ambient. Moreover, the temperature at  $P_{se}$  was strongly resistant to high-frequency power losses, e.g., the temperature amplitude at  $P_{se}$  was only -20 dB for a 10 Hz power loss input. Note that all of the transient thermal impedances were normalized with respect to the steady-state thermal resistance from  $P_{jm}$  to ambient, and the frequency responses of the temperatures at the selected points were normalized by a power loss of 100 W.



Fig. 7.7: FEA model of the general IGBT module.

The accuracy of the proposed frequency-domain analysis method was examined by comparing the estimated transient temperature at  $P_{jm}$ , which was the point most sensitive to power loss changes, with that obtained from the FEA method for an arbitrary loading condition. The estimated temperature at  $P_{jm}$  using the proposed method matched the result of the FEA method very well, as shown in Fig. 7.9. The maximum error was



Fig. 7.8: FEA simulation results for a step-change input power. (a) Normalized transient thermal impedances from the selected points to ambient and(b) frequency responses of the temperatures at *P<sub>jm</sub>*, *P<sub>ce</sub>*, and *P<sub>se</sub>*.

less than 0.3 °C. The estimated transient temperatures at  $P_{ce}$  and  $P_{se}$  are also shown in Fig. 7.9. Quite different thermal responses were observed at the three points: 1) the temperature at  $P_{jm}$  increased and fluctuated significantly, 2) the temperature at  $P_{ce}$ increased and fluctuated slightly, and 3) the temperature at  $P_{se}$  increased slowly with negligible fluctuation. The result was consistent with that the temperature at  $P_{se}$  is resistant to high frequency components of power losses. If the dominant components of the power loss at a loading condition are relatively high-frequency components, the substrate solder is much less stressed.



Fig. 7.9: Comparison of the proposed method with the FEA method. (a) The power loss and the transient temperatures estimated by the two methods at (b)  $P_{jm}$ , (c)  $P_{ce}$ , and (d)  $P_{se}$ .

#### 7.4.3 Frequency Domain Aging Mode Analysis

To study the impact of different real-world loading conditions on the lifetime of IGBT modules, the "general" test IGBT module was used as a switch in the inverter of an HEV under slow acceleration [loading (a)] and fast acceleration and braking [loading (b)] conditions. As bond wires are usually observed as a bottleneck weak point [45], the focus was placed on determining if the device would undergo mixed aging modes under the loading conditions. Thus, the lifetime of the substrate solder was examined under the two loading conditions. The speed of the HEV, frequency spectrum of the generated heat in the IGBT chip, and temperature at  $P_{se}$  under the two loading conditions are shown in Fig. 7.10(a) and Fig. 7.10(b), respectively. The frequency spectrum in the range of 0.1-10 Hzwas used for rough substrate solder aging analysis. The range was selected based on the frequency response of the temperature at  $P_{se}$ . The temperature at  $P_{se}$  was estimated by using the full frequency spectrum to ensure accuracy. The amplitudes of the low frequency components of the heat generated under loading (b) were approximately five times larger than those under loading (a), which indicates that the device undergoes a more stressful thermal cycling under loading (b). As a consequence, the temperature swing of  $P_{se}$  is about 14 °C under loading (b), while only around 3 °C under loading (a). The rainflow counting algorithm [83] was used to process the temperature-time data, i.e., the cyclic information of the temperatures at  $P_{se}$ . Then, by substituting the processed cyclic information of the temperatures at  $P_{se}$  under loadings (a) and (b) into the fatigue lifetime model for the substrate solder [84], the lifetimes of the substrate solder under loadings (a) and (b) were predicted to be 7,442 hours and 1,585 hours, respectively. The lifetime predictions verified that the aging of the substrate solder under loading (a) was



Fig. 7.10: Speeds of HEV, frequency spectra of heat generation, and temperatures of  $P_{se}$  for loading (a) and loading (b).

more severe than that under loading (b). This suggests that the substrate solder can be a bottleneck weak point of the IGBT module, and the IGBT module might undergo mixed aging modes under loading (b).

# 7.5 Summary

This chapter has proposed a computationally efficient, high-fidelity, easily built, online, frequency-domain junction temperature estimation method for IGBT modules. The proposed method characterized the dynamical thermal behavior of an IGBT module by an LTI system, which approximated an FEA thermal model for the IGBT. The junction temperature was then estimated from the FTT-based frequency responses of the LTI system and the heat sources of the IGBT module. The proposed method removed the need for deriving an explicit expression of the system's transfer function and could be easily built using numerical simulations or experiments. The proposed method has been validated by simulation studies for a commercial IGBT module. Results have shown that the accuracy, particularly the accuracy during transient operating conditions, of the proposed method is comparable to the FEA method and is much higher than the commonly used TEC model. However, the computational cost of the proposed method is much lower than that of the FEA method and is comparable to the TEC model.

Further, the proposed method can be extended to estimate the temperatures at weak points in an IGBT module. Simulation results have proved that the temperatures at the weak points can also be accurately estimated. By examining the frequency spectra of the power losses of an IGBT at a loading condition and the frequency responses of the temperatures at weak points, the information on bottleneck weak points under the specific operating conditions can be obtained.

# Chapter 8. Conclusions, Contributions, and Recommendations for Future Research

The goal of this dissertation research was to develop accurate real-time internal temperature estimation and health monitoring methods to provide better overtemperature protection and lifetime prediction for and avoid unexpected wear-out failures of IGBT modules used in many applications, such as automobiles and wind turbines. The thermal dynamics and major aging mechanisms of the "standard" IGBT module were considered in this dissertation research. First, three cost-effective thermal models suitable for realtime implementation were developed to accurately estimate the junction temperature of the IGBTs for better overtemperature protection and lifetime prediction. Second, a physics-of-failure-based practical technique using 2D case temperatures was invented to monitor and identify the major aging processes of the IGBT modules for effective maintenance scheduling. Third, a novel frequency-domain thermal analysis method was developed for fast and accurate estimation of temperature swings at the weak points of the IGBT modules. This method can give a quick determination of the potential bottleneck weak point(s) of the IGBT modules in a specific application, if its typical duty cycle is given and help choose an appropriate lifetime mode for the RUL prediction.

## 8.1 Conclusions and Contributions

Two cost-effective thermal models for healthy IGBT modules were developed and presented in Chapter 3 and Chapter 4. The physics-based improved Cauer-type TEC in Chapter 3 introduced a new concept of lumped-capacitance approximation error, which can be used to determine the number of RC pairs for each layer of the module architecture. Further, an analytical expression of the thermal impedance of the bulky baseplate layer was derived and simplified to a first-order transfer function, which can be represented by an RC pair in a TEC. By using the proposed modeling method, the structure and parameters of a Cauer TEC can be appropriately designed to meet the high-accuracy and low-cost requirements for real-time implementation. A simulation study proved that the proposed Cauer-type TEC model greatly improves the accuracy of the conventional Cauer-type TEC model for the transient junction temperature estimation of IGBT modules with a slightly increased order only.

In Chapter 4, a thermal model based on low-order IIR digital filters was developed to characterize the thermal behaviors of IGBT modules. The coefficients of the IIR digital filters can easily be determined by using some mature digital filter parameter identification techniques, such as the Steiglitz-McBride method. The digital filters can handle dynamics in large frequency range, have stable performance, and model the thermal cross-coupling in the IGBT modules more accurately than TEC models. Simulation results showed that the accuracy of the proposed digital filter thermal model is much higher than the TEC model. One more advantage of the digital IIR filters is that they can be easily implemented in a digital processing system for real-world applications.

In Chapter 5, to properly manage the operation of power converters during the IGBT aging process, an adaptive TEC model was developed that can be used for accurate junction temperature estimation over the lifespan of IGBT modules. Based on a new method for online monitoring of solder cracks in IGBT modules, the changes in the

thermal behavior of the IGBT modules were interpreted by the changes in the EHPP, which is a new concept introduced in this research. Further, based on the EHPP, the changes in the thermal behavior of the IGBT modules were quantified online; and the parameters of the TEC model were updated online as well. Simulation studies validated the effectiveness of using the proposed EHPP and adaptive TEC model to track changes in device thermal resistance from junction to case and achieve improved junction temperature estimation during the process of substrate solder cracks.

All of the three thermal models can be used for real-time junction temperature estimation. The advantage of the improved Cauer-type TEC model is that it has physical meanings. It can incorporate temperature effects on materials' thermal properties and aging effects, such as the substrate solder crack considered in the adaptive thermal model in Chapter 5. However, the improved Cauer-type TEC model is a simplified 1D model. In order to model the thermal behavior of an IGBT module with multiple chips and complex thermal cross-coupling effects, a better choice is the digital filter thermal model. However, since the digital filter thermal model does not have physical meanings, it is not straightforward to include aging effects in the model.

In Chapter 6, multiple aging processes (caused by different mechanisms) of IGBT modules were monitored by only acquiring package surface temperatures at a few selected points. The proposed method was based on the physics-of-failure of the IGBT modules, which was interpreted via abnormal heat generation and heat dissipation path degradation. Further, by taking advantage of the linear thermal behavior of the IGBT modules and introduction of new aging indicators  $k_{hc}$  and  $\alpha_P$ , the proposed method was capable of monitoring the major aging mechanisms and detecting which aging process or

processes occur, which is an advantage over the existing techniques. Also, the implementation of the proposed method has lower hardware complexity and does not affect power converter operation.

In Chapter 7, a novel frequency domain thermal and weak points analysis method was presented. In the first part of the chapter, an online frequency-domain junction temperature estimation method for IGBT modules was proposed. The proposed method characterized the dynamical thermal behavior of an IGBT module by an LTI system. Then, the junction temperature was estimated in the frequency domain by using the frequency responses of the LTI system and FFT-based analysis of the power losses generated in the IGBT module. The proposed method was computationally efficient, had high-fidelity, and was easy to build. In the second part, the bottleneck weak point(s) that limits the lifetime of IGBTs under complex loading conditions was analyzed in the frequency domain. As the aging of the weak points are caused by thermomechanical stresses, the typical frequency responses of temperatures at the weak points of a general IGBT module were analyzed. This revealed that the IGBT power losses at loading conditions with different frequency spectra stress the weak points differently. Therefore, the information on the aging mode(s) of the IGBT can easily be obtained in the frequency domain. The analysis in the frequency domain can help classify the power converter duties in terms of the aging modes of the "standard" IGBT module.

The main contributions of this research are summarized as follows:

1. A comprehensive survey of state-of-the-art thermal models and aging monitoring techniques for IGBT modules was presented. These thermal

models were discussed in terms of accuracy, complexity, and computational cost. Typical drawbacks of the aging monitoring techniques were discussed.

- 2. A new method of constructing an improved low-order Cauer-type TEC model for IGBT modules was proposed. Compared to the conventional Cauer-type TEC, the proposed Cauer-type TEC had a slightly higher order but greatly improved the accuracy of the transient junction temperature estimation for IGBT modules.
- 3. Using low-order IIR digital filters for accurate real-time junction temperature estimation of IGBT modules was proposed. The transient junction temperature of IGBT modules can be accurately estimated by using the IIR filters with the orders even lower than those of the Foster-type TECs.
- 4. An EHPP-based TEC model for IGBT modules, where the parameters have been made adaptive online to substrate solder cracks inside the IGBT modules, was proposed in this research. To the author's knowledge, it is the first time that the aging effects are included in the thermal behavior modeling of IGBT modules.
- 5. A real-time aging monitoring method for IGBT modules using 2D case temperatures was proposed. The proposed method is effective for monitoring major aging processes and identifying their mechanisms in an IGBT module, such as bond wire liftoff, metallization aging, and substrate solder aging. This method is simple to use and easy to implement. To the author's knowledge, this is the first work using surface temperatures for the monitoring of IGBT major aging processes.

- 6. A computationally efficient, high-fidelity, easily built, online, frequencydomain junction temperature estimation method for IGBT modules was proposed. The accuracy of the proposed method is comparable to the FEA method, but the computational cost of the proposed method is comparable to the TEC model.
- 7. The frequency domain thermal analysis method was extended to estimate the temperatures at the weak points of IGBT modules and determine the bottleneck weak point(s) for a given application operation cycle. This may help to classify power converter duty cycles and improve power converter design for specific applications in terms of reliability.

## 8.2 **Recommendations for Future Research**

This dissertation research has developed new cost-effective real-time thermal models that are more accurate than the traditional TEC models, a method for real-time monitoring of the major aging processes, and a weak point aging mode analysis method. These models and methods, however, need to be further improved for more complex scenarios in current or future real-world applications.

As Si IGBT modules usually operate in a temperature range of a few tens of degrees, e.g., 70-125 °C in electric vehicle applications, temperature effects on materials' thermal properties are negligible. The thermal models developed are linear models, which neglect the materials' nonlinearities. However, SiC IGBT may be commercially available in the future. The SiC IGBT-based power converter may operate in a much wider temperature range, as the maximum junction temperature can be 250 °C or higher. For this scenario, it is suggested that research is needed to study the potential impact of the
nonlinearities of the materials' thermal properties on the SiC IGBT device's thermal behavior. Nonlinear thermal models can be built by adding a scheme to update the thermal models' parameters at different device temperatures.

In this dissertation research, an implementation of the health monitoring method uses two thermocouples (one placed at a relatively hot spot and the other placed at a relatively cold spot at the bottom surface of the baseplate) for each switch of the IGBT module. For a 3-phase power inverter, 12 thermocouples will be used. However, the number of thermocouples can be reduced; and their locations can be optimized for a more effective implementation in real-world applications. It is recommended that a sensitivity analysis be conducted of the changes in the values of equivalent thermal resistances to the aging processes of different weak points in an inverter module. Another approach to reducing the cost of temperature sensors is to take advantage of low-cost, large-area printed organic temperature sensors on flexible substrates, e.g., [85]-[88]. An organic temperature sensor array may be placed in the thermal grease between an IGBT module and a cold plate. This approach can realize 2D case temperature sensing in real-world applications and collect enough information for a more detailed analysis of the device's health status.

The weak point analysis in the frequency domain can be advantageous in aging mode analysis with respect to the time-domain cycle counting methods, especially when considering computational costs. Unlike the time-domain methods, the proposed method cannot provide a specific lifetime prediction for each weak point of an IGBT module. Because this method doesn't provide a histogram of temperature cycles, therefore, it cannot use the  $N_f$  cycling curves of the IGBT module. So far, the bottleneck weak point(s) are roughly determined by observing the power loss frequency spectra. It is recommended that, in a future work, damage indicators be extracted from the temperature spectra for each of the weak points to quantifiably compare their severities. If this can be achieved, rules for power converter duty classification in terms of the aging modes of the "standard" IGBT module can be further established. The duty classification can benefit power converter design. As the aging mode of IGBT modules for a typical duty cycle can be known before power converter production, the design of the IGBT modules can be modified to reduce the thermal stress at the bottleneck weak point(s) to prolong the lifetime of the power converter. This can lead to a cost-effective, field-oriented, reliable design for power converters.

# Appendix

Suppose that the *i*th layer is subdivided into  $N_{sub}$  sublayers with the same thickness and is not the thick baseplate layer. Then the heat conduction areas on the top and bottom surfaces,  $A(z_{i,top})$  and  $A(z_{i,bottom})$ , respectively, of the layer are very close, i.e.,

$$A(z_{i,top}) \approx A(z_{i,bottom}) \tag{A.1}$$

Then the lumped-capacitance approximation error of each sublayer,  $CE_{lc,i,sub}$ , can be calculated by multiplying  $c_{v,i}A(z_{i,top})$  by the temperature distribution approximation error of the sublayer, i.e., a triangle shadow area in Fig. A.1.

$$CE_{lc,i,sub} \approx c_{\nu,i}A(z_{i,top}) \cdot \frac{\frac{1}{2N_{sub}}[T(z_{i,top}) - T(z_{i,bottom})]}{T_j - T_c} \cdot \frac{1}{N_{sub}} (z_{i,bottom} - z_{i,top})$$
$$\approx \frac{1}{2N_{sub}^2} \frac{R_{th,i}}{R_{th,jc}} C_{th,i}$$
(A.2)

where the difference between the actual and estimated temperatures at a position is normalized in terms of  $(T_j - T_c)$ . Then,  $CE_{lc,i}$  is the sum of  $CE_{lc,i,sub}$  of all the sublayers,



Fig. A.1: The temperature distribution approximation error of the *i*th layer subdivided into *N* sublayers.

i.e.,  $CE_{lc,i} = N_{sub} \cdot CE_{lc,i,sub}$ , which leads to (3.3).

Let  $Q_i$  be the amount of heat stored in the *i*th layer. Then, the amount of heat stored in the *i*th layer when  $(T_j - T_c)$  is raised by 1 °C, i.e.,  $Q_i/(T_j - T_c)$ , can be calculated as follows:

$$\frac{Q_{i}}{T_{j}-T_{c}} = \int_{z_{i,top}}^{z_{i,bottom}} \frac{T(z)-T_{j}}{T_{j}-T_{c}} c_{v}A(z)dz$$

$$= \int_{z_{i,top}}^{z_{i,bottom}} \frac{T(z_{i,top})-T_{c}}{T_{j}-T_{c}} c_{v}A(z)dz - \int_{z_{i,top}}^{z_{i,bottom}} \frac{T(z_{i,top})-T(z)}{T_{j}-T_{c}} c_{v}A(z)dz$$

$$\approx \frac{\sum_{j=1}^{i} R_{th,j}}{R_{th,jc}} C_{th,i} - \frac{\frac{1}{2}R_{th,j}}{R_{th,jc}} C_{th,i}$$

$$= \frac{1}{R_{th,jc}} C_{th,i} \left( \sum_{j=1}^{i} R_{th,j} - \frac{1}{2} R_{th,i} \right) \qquad (A.3)$$

By adding the heat stored in all the layers, the equation (3.4) for calculating  $C_{th,tot}$  is then obtained.

## References

- J. Ribrant and L. M. Bertling, "Survey of failures in wind power systems with focus on Swedish wind power plants during 1997-2005," *IEEE Trans. Energy Conversion*, vol. 22, no. 1, pp. 167-173, Mar. 2007.
- [2] S. Sheng, "Report on wind turbine subsystem reliability–A survey of various databases," National Renewable Energy Laboratory, June 2013 [Online]. Available: <u>http://www.smartgridinformation.info/pdf/5527\_doc\_1.pdf</u>.
- [3] S. Yang, A. Bryant, P. Mawby, D. Xiang, L. Ran, and P. Tavner, "An industrybased survey of reliability in power electronic converters," *IEEE Trans. Ind. Appl.*, vol. 47, pp. 1441-1451, May/Jun. 2011.
- [4] F. Blaabjerg and K. Ma, "Future on power electronics for wind turbine systems," *IEEE J. Emerg. Sel. Top. Power Electron.*, vol. 1, no. 3, pp. 139–152, Sept. 2013.
- Y. Song and B. Wang, "Survey on reliability of power electronic systems," *IEEE Trans. Power Electron.*, vol. 28, no. 1, pp. 591–604, Jan. 2013.
- [6] Mitsubishi Electric, Appl. Note, *IGBT Modules Application Note*, 2014.
- [7] E. R. Motto, J. F. Donlon, M. Miyazawa, M. Tabata, H. Muraoka, T. Hieda, and T. Radke, "Next generation industrial IGBT module," in *Proc. IEEE Energy Conversion Congress and Exposition*, Sept. 2014, pp. 2893–2896.
- [8] Mitsubishi Electric, "Mitsubishi IGBT module CM400DY-12NF," CM400DY-12NF datasheet, Feb. 2009.
- [9] Dynex Semiconductor, Appl. Note, *IGBT Module Reliability*, AN 5945, 2010.

- [10] M. Ciappa, "Selected failure mechanisms of modern power modules," *Microelectron. Reliab.*, vol. 42, no. 4–5, pp. 653–667, Apr. 2002.
- [11] H. Ye, M. Lin, and C. Basaran, "Failure modes and FEM analysis of power electronic packaging," *Finite Elem. Anal. Des.*, vol. 38, no. 7, pp. 601–612, May 2002.
- [12] B. Ji, X. Song, E. Sciberras, W. Cao, Y. Hu, and V. Pickert, "Multi-objective design of IGBT power modules considering power cycling and thermal cycling," *IEEE Trans. Power Electron.*, vol. 30, no. 5, pp. 2493–2504, May 2015.
- [13] Y. Celnikier, L. Dupont, E. Hervé, G. Coquery, and L. Benabou, "Optimization of wire connections design for power electronics," *Microelectron. Reliab.*, vol. 51, no. 9–11, pp. 1892–1897, Sept. 2011.
- [14] S. Kraft, A. Schletz, and M. März, "Reliability of silver sintering on DBC and DBA substrates for power electronic applications," in *Proc. 7th International Conference on Integrated Power Electronics Systems*, Mar. 2012, pp. 1–6.
- [15] D. A. Murdock, J. E. R. Torres, J. J. Connors, and R. D. Lorenz, "Active thermal control of power electronic modules," *IEEE Trans. Ind. Appl.*, vol. 42, no. 2, pp. 552–558, Mar. 2006.
- [16] A. Morozumi, K. Yamada, T. Miyasaka, S. Sumi, and Y. Seki, "Reliability of power cycling for IGBT power semiconductor modules," *IEEE Trans. Ind. Appl.*, vol. 39, no. 3, pp. 665–671, 2003.
- [17] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. D. Doncker, Semiconductor Power Devices: Physics, Characteristics, Reliability. Berlin, Germany: Springer, 2011.

- [18] V. Smet, F. Forest, J. Huselstein, F. Richardeau; Z. Khatir, S. Lefebvre, and M. Berkani, "Ageing and failure modes of IGBT modules in high-temperature power cycling," *IEEE Trans. Ind. Electron.*, vol. 58, no. 10, pp. 4931–4941, Oct. 2011.
- [19] P. Ghimire, S. Beczkowski, S. Munk-Nielsen, B. Rannestad, and P. B. Thogersen, "A review on real time physical measurement techniques and their attempt to predict wear-out status of IGBT," in *Proc. 15th European Conf. Power Electron. and Appl.*, Sept. 2013, pp. 1–8.
- [20] Infineon Technologies, Appl. Note, Calculation of major IGBT operating parameters, ANIP9931E, 1999.
- [21] K. Ma, A. S. Bahman, S. Beczkowski, and F. Blaabjerg, "Complete loss and thermal model of power semiconductors including device rating information," *IEEE Trans. Power Electron.*, vol. 30, no. 5, pp. 2556-2569, May 2015.
- [22] A. Castellazzi, "Comprehensive compact models for the circuit simulation of multichip power modules," *IEEE Trans. Power Electron.*, vol. 25, no. 5, pp. 1251– 1264, May 2010.
- [23] M. Musallam and C. M. Johnson, "Real-time compact thermal models for health management of power electronics," *IEEE Trans. Power Electron.*, vol. 25, no. 6, pp. 1416–1425, Jun. 2010.
- [24] P. L. Evans, A. Castellazzi, and C. M. Johnson, "Automated fast extraction of compact thermal models for power electronic modules," *IEEE Trans. Power Electron.*, vol. 28, no. 10, pp. 4791–4802, Oct. 2013.

- [25] Z. Wang, W. Qiao, and L. Qu, "Frequency-domain transient temperature estimation and aging analysis for weak points of IGBT modules," in *Proc. IEEE Energy Conversion Congress and Exposition*, Sept. 2014, pp. 4036-4042.
- [26] F. Masana, "A new approach to the dynamic thermal modelling of semiconductor packages," *Microelectron. Reliab.*, vol. 41, no. 6, pp. 901–912, Jun. 2001.
- [27] T. Gachovska, B. Tian, J. Hudgins, W. Qiao, and J. Donlon, "A real-time thermal model for monitoring of power semiconductor devices," *IEEE Trans. Ind. Appl.*, vol. 51, no. 4, pp. 3361–3367, Jul./Aug. 2015.
- [28] R. Wu, H. Wang, K. Ma, P. Ghimire, F. Iannuzzo, and F. Blaabjerg, "A temperature-dependent thermal model of IGBT modules suitable for circuit-level simulations," in *Proc. IEEE Energy Conversion Congress and Exposition*, Sept. 2014, pp. 2901-2908.
- [29] V. Székely and T. V. Bien, "Fine structure of heat flow path in semiconductor devices: A measurement and identification method," *Solid-State Electron.*, vol. 31, pp. 1363–1368, 1988.
- [30] V. Szekely and M. Rencz, "Thermal dynamics and the time constant domain," *IEEE Trans. Components Packag. Technol.*, vol. 23, no. 3, pp. 587–594, Sept. 2000.
- [31] F. Masana, "A new approach to the dynamic thermal modelling of semiconductor packages," *Microelectron. Reliab.*, vol. 41, no. 6, pp. 901–912, Jun. 2001.
- [32] P. Malberti, M. Ciappa, and W. Fichtner, "Thermal component model for electrothermal analysis of IGBT module systems," *IEEE Trans. Adv. Packag.*, vol. 24, no. 3, pp. 401–406, Aug. 2001.

- [33] C. Batard, N. Ginot, and J. Antonios, "Lumped dynamic electrothermal model of IGBT module of inverters," *IEEE Trans. Components, Packag. Manuf. Technol.*, vol. 5, no. 3, pp. 355–364, Mar. 2015.
- [34] M. Janicki, G. De Mey, and A. Napieralski, "Transient thermal analysis of multilayered structures using Green's functions," *Microelectron. Reliab.*, vol. 42, no. 7, pp. 1059–1064, Jul. 2002.
- [35] B. Du, J. L. Hudgins, E. Santi, A. T. Bryant, P. R. Palmer, and H. A. Mantooth, "Transient electrothermal simulation of power semiconductor devices," *IEEE Trans. Power Electron.*, vol. 25, no. 1, pp. 237–248, Jan. 2010.
- [36] K. Cole, A. Haji-Sheikh, J. Beck, and B. Litkouhi, *Heat Conduction Using Green's Functions*, 2<sup>nd</sup> ed. New York: CRC Press, 2011.
- [37] K. A. Woodbury and J. V. Beck, "Heat conduction in a planar slab with power-law and polynomial heat flux input," *Int. J. Therm. Sci.*, vol. 71, pp. 237–248, Sept. 2013.
- [38] F. de Monte, "Multi-layer transient heat conduction using transition time scales," *Int. J. Therm. Sci.*, vol. 45, no. 9, pp. 882–892, Sept. 2006.
- [39] J. J. Nelson, G. Venkataramanan, and A. M. EL-Refaie, "Fast thermal profiling of power semiconductor devices using Fourier techniques," *IEEE Trans. Ind. Electron.*, vol. 53, no. 2, pp. 521–529, Apr. 2006.
- [40] N. Rinaldi, "On the modeling of the transient thermal behavior of semiconductor devices," *IEEE Trans. Electron Devices*, vol. 48, no. 12, pp. 2796–2802, Dec. 2001.
- [41] M. Touzelbaev, J. Miler, Y. Yang, G. Refai-Ahmed, and K. Goodson, "Highefficiency transient temperature calculations for applications in dynamic thermal

management of electronic devices," J. Electron. Packag., vol. 135, no. 3, pp. 031001-1-031001-8, 2013.

- [42] J. N. Davidson, D. A. Stone, and M. P. Foster, "Real-time prediction of power electronic device temperatures using PRBS-generated frequency-domain thermal cross coupling characteristics," *IEEE Trans. Power Electron.*, vol. 30, no. 6, pp. 2950–2961, Jun. 2015.
- [43] J. Lutz, "IGBT-modules: Design for reliability," in Proc. 13th European Conference on Power Electronics and Applications, Sept. 2009, pp. 1–3.
- [44] P. Ghimire, S. Beczkowski, S. Munk-Nielsen, B. Rannestad, and P. B. Thogersen,
  "A review on real time physical measurement techniques and their attempt to predict wear-out status of IGBT," in *Proc. 15th European Conf. Power Electron.* and Appl., Sept. 2013, pp. 1–8.
- [45] B. Ji, V. Pickert, W. Cao, and B. Zahawi, "In-situ diagnostics and prognostics of wire bonding faults in IGBT modules for electric vehicle drives," *IEEE Trans. Power Electron.*, vol. 28, no. 12, pp. 5568–5577, Dec. 2013.
- [46] V. Smet, F. Forest, J. Husselstein, A. Rashed, and F. Richardeau, "Evaluation of Vce monitoring as a real-time method to estimate ageing of bond wire-IGBT modules stressed by power cycling," *IEEE Trans. Ind. Electron.*, vol. 60, no. 7, pp. 2760–2770, Jul. 2013.
- [47] B. Ji, X. Song, W. Cao, V. Pickert, Y. Hu, J. Mackersie, and G. Pierce, "In-situ diagnostics and prognostics of solder fatigue in IGBT modules for electric vehicle drives," *IEEE Trans. Power Electron.*, vol. 30, no. 3, pp. 1535-1543, Mar. 2015.

- [48] Y. Xiong, X. Cheng, Z. J. Shen, C. Mi, H. Wu, and V. K. Garg, "Prognostic and warning system for power-electronic modules in electric, hybrid electric, and fuelcell vehicles," *IEEE Trans. Ind. Electron.*, vol. 55, no. 6, pp. 2268–2276, Jun. 2008.
- [49] Y. Avenas, L. Dupont, and Z. Khatir, "Temperature measurement of power semiconductor devices by thermo-sensitive electrical parameters—A review," *IEEE Trans. Power Electron.*, vol. 27, no. 6, pp. 3081–3092, Jun. 2012.
- [50] S. Zhou, L. Zhou, and P. Sun, "Monitoring potential defects in an IGBT module based on dynamic changes of the gate current," *IEEE Trans. Power Electron.*, vol. 28, no. 3, pp. 1479–1487, Mar. 2013.
- [51] D. Xiang, L. Ran, P. Tavner, S. Yang, A. Bryant, and P. Mawby, "Condition monitoring power module solder fatigue using inverter harmonic identification," *IEEE Trans. Power Electron.*, vol. 27, no. 1, pp. 235–247, Jan. 2012.
- [52] D. Brown, M. Abbas, A. Ginart, I. Ali, P. Kalgren, and G. Vachtsevanos, "Turn-off time as an early indicator of insulated gate bipolar transistor latch-up," *IEEE Trans. Power Electron.*, vol. 27, no. 2, pp. 479–489, Feb. 2012.
- [53] D. Xiang, L. Ran, and P. Tavner, "Monitoring solder fatigue in a power module using case-above-ambient temperature rise," *IEEE Trans. Ind. Appl.*, vol. 47, no. 6, pp. 2578–2591, Nov. 2011.
- [54] Z. Wang and W. Qiao, "A Physics-Based Improved Cauer-Type Thermal Equivalent Circuit for IGBT Modules," *IEEE Trans. Power Electron.*, vol. 31, no. 10, pp. 6781-6786, Oct. 2016.
- [55] B. Vermeersch and G. De Mey, "A fixed-angle heat spreading model for dynamic thermal characterization of rear-cooled substrates," in *Proc. 23rd Annual IEEE*

Semiconductor Thermal Measurement and Management Symposium, Mar. 2007, pp. 95–101.

- [56] Z. Wang and W. Qiao, "An online frequency-domain junction temperature estimation method for IGBT modules," *IEEE Trans. Power Electronics*, vol. 30, no. 9, pp. 4633-4637, Aug. 2015.
- [57] Z. Wang and W. Qiao, "Real-Time Junction Temperature Estimation for IGBT Modules Using Low-Order Digital Filters," in *Proc. IEEE Power Electronics and Motion Control Conference (IPEMC-ECCE Asia)*, May 2016, pp. 3398-3402.
- [58] MathWorks, "IIR Filter Design," 2015 [Online]. Available: http://www.mathworks.com/help/signal/ug/iir-filter-design.html.
- [59] K. Steiglitz and L. E. McBride, "A technique for the identification of linear systems," *IEEE Trans. Automat. Contr.*, vol. 10, no. 4, pp. 461-464, Oct. 1965.
- [60] P. Stoica and T. Soderstrom, "The Steiglitz-McBride identification algorithm revisited-convergence analysis and accuracy aspects," *IEEE Trans. Automat. Contr.*, vol. 26, no. 3, pp. 712–717, Jun. 1981.
- [61] Z. Wang, W. Qiao, B. Tian, and L. Qu, "An effective heat propagation path-based online adaptive thermal model for IGBT modules," in *Proc. IEEE Appl. Power Elec. Conf. and Expo.*, Mar. 2014, pp. 513-518.
- [62] Infineon Technologies, Appl. Note, *Calculation of major IGBT operating* parameters, ANIP9931E, 1999.
- [63] C. Busca, R. Teodorescu, F. Blaabjerg, S. Munk-Nielsen, L. Helle, T. Abeyasekera, and P. Rodriguez, "An overview of the reliability prediction related aspects of high

power IGBTs in wind power applications," *Microelectron. Reliab.*, vol. 51, no. 9–11, pp. 1903–1907, Sept. 2011.

- [64] S. Yang, D. Xiang, A. Bryant, P. Mawby, L. Ran, and P. Tavner, "Condition monitoring for device reliability in power electronic converters: A review," *IEEE Trans. Power Electron.*, vol. 25, no. 11, pp. 2734–2752, Nov. 2010.
- [65] OMEGA Engineering Inc., Technical Temperature Reference [Online]. Available: <u>http://www.omega.com/temperature/z/zsection.asp.</u>
- [66] Maxim Integrated, "Cold-Junction Compensated Thermocouple-to-Digital Converter," MAX31855 datasheet, Jan. 2015.
- [67] X. Perpina, X. Jordà, M. Vellvehi, J. Rebollo, and M. Mermet-Guyennet "Longterm reliability of railway power inverters cooled by heat-pipe-based systems," *IEEE Trans. Ind. Electron.*, vol. 58, no. 7, pp. 2662–2672, Jul. 2011.
- [68] R. Skuriat, J. F. Li, P. A. Agyakwa, N. Mattey, P. Evans, and C. M. Johnson, "Degradation of thermal interface materials for high-temperature power electronics applications," *Microelectron. Reliab.*, vol. 53, no. 12, pp. 1–10, Jun. 2013.
- [69] M. Schulz, S. Allen, and W. Pohl, "The crucial influence of thermal interface material in power electronic design," in *Proc. 29<sup>th</sup> Annual IEEE Semiconductor Thermal Measurement and Management Symposium*, Mar. 2013, pp. 251–254.
- [70] Fuji Electric, Appl. Note, Measurement of the Case Temperature, REH983a, 2004.
- [71] E. E. Kostandyan and J. D. Sorensen, "Reliability of wind turbine components– Solder elements fatigue failure," in *Proc. Annual Reliability and Maintainability Symposium*, Jan. 2012, pp. 1–7.

- [72] Y. Nishimura, K. Oonishi, A. Morozumi, E. Mochizuki, and Y. Takahashi, "All lead free IGBT module with excellent reliability," in *Proc. 17th International Symposium on Power Semiconductor Devices and ICs*, May 2005, pp. 79–82.
- [73] J. P. Sommer, T. Licht, H. Berg, K. Appelhoff, and B. Michel, "Solder fatigue at high-power IGBT modules," in *Proc. 4th International Conference on Integrated Power Systems*, Jun. 2006, pp. 1–6.
- [74] C. J. M. Lasance, "How to estimate heat spreading effects in practice," J of Electronic Packaging, vol. 132, no. 3, pp. 031004-1-031004-7, Sept. 2010.
- [75] A. Jamnia, Practical Guide to the Packaging of Electronics, New York: Marcel Dekker, Inc. 2003.
- [76] P. Kawka, G. De Mey, and B. Vermeersch, "Thermal characterization of electronic packages using the Nyquist plot of the thermal impedance," *IEEE Trans. Components Packag. Technol.*, vol. 30, no. 4, pp. 660–665, Dec. 2007.
- [77] L. Wei, J. Mcguire, and R. A. Lukaszewski, "Analysis of PWM frequency control to improve the lifetime of PWM inverter," *IEEE Trans. Ind. Appl.*, vol. 47, no. 2, pp. 922–929, Mar./Apr. 2011.
- [78] Z. Wang, B. Tian, W. Qiao, and L. Qu, "Real-Time Aging Monitoring for IGBT Modules Using Case Temperature," *IEEE Trans. Ind. Electron.*, vol. 63, no. 2, pp. 1168-1178, Feb. 2016.
- [79] Y. Chen and X. Wu, "A comprehensive analytical and experimental investigation of wire bond life for IGBT modules," in *Proc. 27th Annual Applied Power Electronics Conference and Exposition*, Feb. 2012, pp. 2298–2304.

- [80] M. Ciappa, "Lifetime prediction on the base of mission profiles," *Microelectron. Reliab.*, vol. 45, no. 9–11, pp. 1293–1298, Sept. 2005.
- [81] Powerex, Appl. Note, Normalized Transient Thermal Impedance (Z<sub>th</sub>) Using T<sub>c</sub> Reference Point Under Chip for 600 V and 1200 V A, NF, NFH and S Series IGBT Modules, 2012.
- [82] T. Poller and S. D'Arco, "Influence of thermal cross-couplings on power cycling lifetime of IGBT power modules," in *Proc. 7th International Conference on Integrated Power Electronics Systems*, Mar. 2012, pp. 1-6.
- [83] Standard Practices for Cycle Counting in Fatigue Analysis, ASTM E1049-85(2005), 2005.
- [84] M. Musallam, C. M. Johnson, and C. Bailey, "In-service life consumption estimation in power modules," in *Proc. 13th International Conference on Power Electronics and Motion Control*, Sept. 2008, pp. 76–83.
- [85] W.-P. Shih, L.-C. Tsao, C.-W. Lee, M.-Y. Cheng, C. Chang, Y.-J. Yang, and K.-C. Fan, "Flexible temperature sensor array based on a graphite-polydimethylsiloxane composite," *Sensors (Basel).*, vol. 10, no. 4, pp. 3597–3610, Jan. 2010.
- [86] A. Aliane, V. Fischer, M. Galliari, L. Tournon, R. Gwoziecki, C. Serbutoviez, I. Chartier, and R. Coppard, "Enhanced printed temperature sensors on flexible substrate," *Microelectronics J.*, vol. 45, no. 12, pp. 1621–1626, Dec. 2014.
- [87] X. Ren, P. K. L. Chan, J. Lu, B. Huang, and D. C. W. Leung, "High dynamic range organic temperature sensor.," *Adv. Mater.*, vol. 25, no. 9, pp. 1291–5, Mar. 2013.

[88] T. Q. Trung, S. Ramasundaram, B.-U. Hwang, and N.-E. Lee, "An all-elastomeric transparent and stretchable temperature sensor for body-attachable wearable electronics," *Adv. Mater.*, vol. 28, no. 3, pp. 502-509, 2015.

### **List of Publications and Patent**

#### **Referred Journal Papers**

- [1] Z. Wang and W. Qiao, "A Physics-Based Improved Cauer-Type Thermal Equivalent Circuit for IGBT Modules," *IEEE Trans. Power Electron.*, vol. 31, no. 10, pp. 6781-6786, Oct. 2016.
- [2] Z. Wang, B. Tian, W. Qiao, and L. Qu, "Real-Time Aging Monitoring for IGBT Modules Using Case Temperature," *IEEE Trans. Ind. Electron.*, vol. 63, no. 2, pp. 1168-1178, Feb. 2016.
- [3] Z. Wang and W. Qiao, "An Online Frequency-Domain Junction Temperature Estimation Method for IGBT Modules," *IEEE Trans. Power Electron.*, vol. 30, no. 9, pp. 4633-4637, Sept. 2015.

#### **Refereed Conference Proceeding Papers**

- [1] Z. Wang and W. Qiao, "Real-Time Junction Temperature Estimation for IGBT Modules Using Low-Order Digital Filters," in *Proc. IEEE Power Electronics and Motion Control Conference (IPEMC-ECCE Asia)*, May 2016, pp. 3398-3402.
- [2] Z. Wang, W. Qiao, and L. Qu, "Frequency-Domain Transient Temperature Estimation and Aging Analysis for Weak Points of IGBT Modules," in *Proc. IEEE Energy Conversion Congress and Exposition*, Sep. 2014, pp. 4036-4042.
- [3] Z. Wang, W. Qiao, B. Tian, and L. Qu, "An Effective Heat Propagation Path-based Online Adaptive Thermal Model for IGBT Modules," *Proc. IEEE Applied Power Electronics Conference and Exposition*, Mar. 2014, pp. 513-518.

- [4] B. Tian, Z. Wang, W. Qiao, and L. Qu, "Study on Case Temperature Distribution for Condition Monitoring of Multidie IGBT Modules," in *Proc. IEEE Applied Power Electronics Conference and Exposition*, Mar. 2014, pp. 2564-2568.
- [5] B. Tian, W. Qiao, Z. Wang, T. Gachovska, and J. L. Hudgins, "Monitoring IGBTs Health Condition via Junction Temperature Variations," in *Proc. IEEE Applied Power Electronics Conference and Exposition*, Mar. 2014, pp. 2550-2555.

#### Patent

 W. Qiao, Z. Wang, and L. Qu, "Monitoring Aging of Power Semiconductor Devices Based on Case Temperature," U.S. Patent Application No. 62/320008.